# The influence of overseas study and work experience on corporate environmental disclosures: evidence from Vietnam

Hang Ngoc My Le

University of Economics, Hue University, Hue City, Vietnam and School of Accounting, Information Systems and Supply Chain, RMIT University, Melbourne, Australia, and

Brendan Thomas O'Connell and Maryam Safari School of Accounting, Information Systems and Supply Chain, RMIT University, Melbourne, Australia

### Abstract

**Purpose** – Drawing from Upper-Echelons Theory (UET), this paper aims to examine whether an increasing number of board members studying and working overseas, especially in Anglo countries, provides some impetus for increased corporate environmental disclosures (CED) in Vietnam.

**Design/methodology/approach** – This study used quantitative data collection and analysis. The data collection involved a content analysis of annual, sustainability and integrated reports to capture the quality and quantity of CED. The authors subsequently developed ordered probit models to quantitatively test the hypotheses.

**Findings** – The authors find that board members studying in Anglo countries positively impact firms' levels of CED in emerging economies. However, overseas work experience is found to be an insignificant explanatory variable. Further, the findings suggest that, in Vietnam, Chairs appear to be more influential than chief executive officers in affecting CED levels.

**Practical implications** – Despite the positive influence of overseas study, the authors find overall levels of CED in Vietnam remain relatively low. This suggests the necessity of dialogue about potential reform in CED policies, which could involve the introduction of mandatory reporting requirements. In addition, to enhance sustainability disclosures, shareholders should appoint board members who possess international qualifications.

**Originality/value** – This study adds to the literature exploring the impacts of Anglo cultural traits of board members on CED levels, within an economy transitioning from a communist ideology to a market-oriented system context. The connection between international study and cultural norms, beliefs and traditions in these countries and their positive influence on directors' values and attitudes towards CED have not yet been studied. The study also extends UET by examining the potential positive influence of different national contexts on board members' education levels.

**Keywords** Corporate governance, Transitional economy, Board backgrounds, Corporate environmental disclosures, Overseas study, Overseas work experience, Upper-Echelons theory

Paper type Research paper

The authors thank Professor John Dumay, Department of Accounting & Corporate Governance, Macquarie University, Australia for his comments on an earlier draft of the paper. The authors also offer sincere gratitude to the guest editor and anonymous reviewers of the journal for their constructive comments and feedback.

Corporate environmental disclosures

Received 30 November 2020 Revised 20 April 2021 12 July 2021 Accepted 17 August 2021



Meditari Accountancy Research © Emerald Publishing Limited 2049-372X DOI 10.1108/MEDAR-11-2020-1109

### MEDAR 1. Introduction

As business activities are significant contributors to global climate change and global warming, it is evident companies should place greater focus on corporate environmental disclosures (CED) (Lee, 2017; Lewis *et al.*, 2014). Many stakeholders expect information about how firm activities affect the environment (Rupley *et al.*, 2012). Further, firms gain a competitive advantage when information about corporate environmental performance is disclosed, as this may lessen stakeholder uncertainty about the company (Simnett *et al.*, 2009).

Despite the well-documented positive value of CED [1], levels of CED in emerging countries remain low when compared to developed countries (Cui *et al.*, 2020; Welbeck, 2017). A reason for this gap may be that many emerging countries possess weak corporate governance (CG) mechanisms, leading to adverse outcomes, such as low disclosure levels (Mahmood *et al.*, 2019; Seibert and Macagnan, 2019). Moreover, environmental regulations in emerging countries are often deficient or weakly enforced (Mahmood *et al.*, 2019). Public awareness of environmental impacts is also relatively low in these countries (Lu and Abeysekera, 2015; Situ and Tilt, 2018). Consequently, studies into the drivers of CED are essential to understanding the distinct context of emerging countries and ultimately to increase their levels of transparency.

The influence of culture, as a set of societal values explaining and determining differences in cross-national accounting and disclosure practices, has been subjected to a wealth of research (Khlif *et al.*, 2015). Similarly, the importance of the level of education amongst the board of directors (BoDs) has received some attention (Katmon *et al.*, 2017; Kipngetich *et al.*, 2019). However, there is little evidence around the cultural traits of board members (such as a Confucian orientation) and the impact they may have on sustainability reporting (Brochet *et al.*, 2019; Cui *et al.*, 2020). According to Upper-Echelons Theory (UET), board member qualifications are indicators of top management's personal and cognitive values and influence strategic choices (Hambrick and Mason, 1984). Cultural norms, beliefs and traditions may intertwine with overseas study to influence directors' attitudes towards CED.

This paper, therefore, aims to examine whether increasing numbers of board members studying and working overseas, especially in Anglo countries [2], provide the impetus for increased CED in Vietnam. In doing so, this study uses quantitative data collection and analysis, which involves a content analysis of annual, sustainability and integrated reports to capture the quality and quantity of CED. Ordered probit models are used to test the potential impacts of board members' overseas study (*BMOS*) and work experience (*BMOW*) on CED. The level of CED is measured through a set of environmental items based on Global Reporting Initiative (GRI) guidelines in the context of Vietnam.

Our study makes seven noteworthy contributions to theory and the literature. Firstly, our research extends UET by examining an overseas study and work experience; two potential contributors to board members' personal and cognitive values. Prior UET studies that have examined educational background have focussed on the type of education such as possession of an Master of Business Administration (MBA) or an accounting/finance or law degree (Bamber *et al.*, 2010; Ge *et al.*, 2011) or level of education such as possession of a Bachelors or Masters degree (Qi *et al.*, 2018). These studies examine accounting choices and none have considered the *source* of that education. Moreover, there has been a dearth of studies to date that have examined the potential impact of backgrounds on non-financial strategic choices such as environmental issues. Where environmental issues have been analysed, researchers such as Aguilera *et al.* (2007) and Jackson and Apostolakou (2010) have mainly focussed on environmental practices and performance. Studies into factors driving environmental disclosures (Clarkson *et al.*, 2008; Lewis *et al.*, 2014) have not

examined the source of education of firms' leadership. Given that exposure to another country that is culturally distant may potentially alter a leader's field of vision, selective perception and interpretation of environmental concerns, this gap is worthy of investigation.

Secondly, there are few studies on environmental issues using non-US samples. An exception is Katmon *et al.* (2017), who examined the relationship between board diversity and corporate social responsibility (CSR) quality in Malaysia but they used a resource-based view theory of the firm rather than a UET lens. As the heterogeneity of firm leadership differs globally, it is not clear that UET predictions are globally valid. Western leadership usually possesses a large amount of discretion encouraged by risk-taking investors, strong societal beliefs in individualism and correspondingly relaxed institutional environments (Hambrick, 2007). Countries such as Vietnam place greater weight on the importance of collectivism and businesses face heavy government intervention and oversight into their affairs. These environmental differences underscore the importance of extending the scope of analysis of UET in voluntary reporting beyond US borders.

Thirdly, our research contributes to the literature by highlighting the value of exposure to Anglo cultural influences on board members' awareness of CED, especially in the unique context of a country transitioning from a communist or socialist ideology system to a market-oriented system. Fourthly, extending prior studies such as Liao et al. (2020), Xixiong et al. (2018) and Slater and Dixon-Fowler (2009), our study further investigates the cultural traits of board members at multiple levels. This includes boards in aggregate and comparing the chief executive officer (CEOs') influence with that of Chairs on CED levels. Fifthly, in contrast to prior studies that found a global standardisation (e.g. GRI) of sustainability reporting (Abeydeera et al., 2016; Ismaeel and Zakaria, 2019; Welbeck, 2017), we found the significant influence on CED of exposure to regional cultures through overseas education. Sixthly, by demonstrating the distinct characteristics of board members' overseas education, our research also extends prior studies, such as Ismaeel and Zakaria (2019), Abevdeera et al. (2016); Khalil and O'Sullivan (2017) and Welbeck (2017), to State-led transition economies, such as Vietnam. Seventhly, our study addressed a major criticism of Hofstede's cultural dimensions by previous studies such as Khlif (2016) and López-Duarte et al. (2016) by using 10 regional cultural clusters as defined by House (2004) based on cultural values, country economic development and linguistic nations across administrative borders.

This paper proceeds as follows: Section 2 explains the Vietnamese research context, including the increasing numbers of Vietnamese people studying overseas and a rising cultural influence from Anglo countries, which are two significant outcomes resulting from Vietnam's economic reforms and its colonial past. Section 3 reviews prior studies and describes the theoretical framework for this study and then proposes research hypotheses linked to the key research questions. Section 4 provides the research methodology and Section 5 presents the results of the study, discussion and possible explanations for the findings. Finally, Section 6 provides the conclusion, including research contributions, the potential implications of these findings, study limitations and avenues for future research.

### 2. Institutional setting

Vietnam provides a unique context for the present study because of its distinct historical and cultural characteristics. Over the modern era, Vietnam has been characterised by a weak CG environment, low public awareness of environmental impacts and significant environmental challenges (Dao and Ofori, 2010). Although Vietnam has some regulations regarding corporate environmental activities, such as Circular 25/2019/TT-BTNMT [3],

Decree 40/2019/ND-CP [4] and Decree 155 [5], these regulations seem to be fewer in comparison with developed countries and also less effective in general (Chu, 2018; Clarke and Vu, 2021; Hoang *et al.*, 2015). In Vietnam, the impact of environmental laws on changing corporate behaviours, in terms of environmental protection and reporting, has been limited (Hoang *et al.*, 2015). Moreover, it should be noted that until now, environmental disclosures in Vietnam have been voluntary, although guidelines have been issued. The GRI guidelines are also voluntary in Vietnam.

Meanwhile, in Anglo countries, a combination of voluntary and mandatory reporting of CED occurs. For example, in Australia, heavy emitters of carbon dioxide are required to annually report their emissions to the government regulator under the National Greenhouse and Energy Reporting (NGER) Act (NGER, 2019) [6]. No such requirements currently exist within Vietnam. Moreover, Vietnam's political system, economy and culture are starkly different from other emerging countries because of its communist and colonial history (Logevall, 2012). Vietnam is the largest socialist country in Southeast Asia and this socialist influence may affect CED given the central government's tendency to oversee most aspects of corporate operations (Benedict, 2019).

Since major economic reforms in 1986, the Communist Party of Vietnam has enacted an "open-door policy" to the international economy. However, two conflicting ideas have emerged around the value of Vietnam undertaking economic reforms. The first view came from Vietnamese people who were influenced by Anglo culture via their overseas exposure. This group argued that economic reforms were necessary because international organisations such as the World Bank and the World Trade Organisation (WTO) enhanced economic, legal and institutional reforms, and therefore, created new positive influences, particularly for the private sector (Michalopoulos, 1999). In contrast, the second view, that economic reforms were unnecessary, tended to be held by many traditional Vietnamese people who had not been exposed to the outside world and so were sceptical of the potential opportunities afforded to Vietnam by opening up its economy and joining the WTO (Pham, 2013). These insiders also thought Vietnam's reforms were a consequence of crises, including the collapse of the centrally-planned economy in 1980, the Asian financial crisis in the 1990s and ineffective economic management (Pham, 2013). These conflicting views around reform provided significant issues for Vietnamese international alumni returning home and working in sectors or institutions dominated by Vietnamese conservatives (Pham, 2019c). These conflicts led to a number of challenges that occurred in the movement from a controlled economy to a market model with a communist political system and one-party rule. For example, conservatives, particularly in state-owned enterprises, who were struggling to compete in a market economy, resisted moves towards economic liberalism (Ashwill and Thai, 2005; Pham, 2013).

Two significant outcomes that resulted from Vietnam's economic reforms and its colonial past are the increasing numbers of Vietnamese people studying overseas and a rising cultural influence from Anglo countries (Oliver and Nguyen, 2010). Further, immense immigration waves of Vietnamese people fleeing Communist rule to developed nations, such as the USA, Canada, Australia and the UK from 1975 to 1995, contributed to a significant Western impact on the education of Vietnam's people and economy. Over the past 20 years or so, these immigrants have also acted as a base for their Vietnamese relatives to "come across" and study in Anglo countries (Hoang, 2010; Parsons and Vézina, 2018). Thus, the increasing numbers of Vietnamese people studying overseas, especially in Anglo countries, provide an impetus for increased CED when they return to their homeland.

### 3. Literature review and hypothesis development

3.1 Qualifications [7], work experience and corporate environmental disclosures A wealth of knowledge exists on the influence of top management characteristics, such as independence, duality, gender, ownership and international influences, on disclosures (Ben-Amar *et al.*, 2017; Cui *et al.*, 2020; Ismaeel and Zakaria, 2019; Xiao and Yuan, 2007). As more studies of CED are conducted, a greater range of new board features and different outcomes are explored (Ismaeel and Zakaria, 2019; Muttakin *et al.*, 2018). However, to date, very little research has examined the association between top management cultural-oriented characteristics and disclosures (Brochet *et al.*, 2019; Zmić *et al.*, 2020, for a review). Specifically, there has been a lack of studies analysing how overseas education and work experience may influence executive decision-making, especially regarding sustainability reporting. Exposure to different countries and perspectives may be critical to executives' attitudes towards non-financial disclosures.

Although the overseas study is more likely to be significant because of Vietnamese economic reforms and historical characteristics (Oliver and Nguyen, 2010), its impacts on CED have not yet been examined. Zrnić *et al.* (2020) suggest that future research should analyse the influence of directors' education levels on sustainability reporting. Previous studies, for example, Katmon *et al.* (2017), Lewis *et al.* (2014), Ma *et al.* (2019) and Kipngetich *et al.* (2019), limited their examinations to local education levels of board members. Meanwhile, several studies, such as Liao *et al.* (2020), Xixiong *et al.* (2018) and Slater and Dixon-Fowler (2009), found a positive relationship between executives' international backgrounds and CSR performance but did not examine board members' education levels or examine sustainability reporting. This research gap is significant as, in a global context, overseas qualifications and work experience obtained by board members are viewed as a valuable form of an intangible asset (Barney, 1992; Carpenter *et al.*, 2001). Top managers with overseas backgrounds may generate significant value for their firms through their distinctive management strategies leading to better firm performance (Slater and Dixon-Fowler, 2009).

The present study uses UET (Hambrick and Mason, 1984) to provide a comprehensive explanation for the influence of board members' overseas qualifications and work experience on environmental reporting. UET provides important insight into the effects of directors' characteristics on firms' strategic choices to innovate, which, in turn, impact firm outcomes (Hambrick, 2007; Hambrick and Mason, 1984). Hambrick (2007) explains that UET has two interconnected parts:

- (1) Executives act on the basis of their personalised interpretations of the strategic situations they face.
- (2) These personalised understandings are influenced by executives' experiences, values and personalities.

This theory posits that to understand the motivation behind firms' actions requires us to consider the biases and dispositions of firms' leadership. Hambrick (2007) argues that a focus on the characteristics of the top management team will provide more robust explanations of firm outcomes than will a focus on individuals (e.g. CEO) as leadership is a shared activity. He also contends that the demographic characteristics of leaders can be used as valid, albeit incomplete and imprecise, proxies of their cognitive frames. Specifically, researchers can reliably use the information on executives' functional backgrounds, industry and firm tenures, educational credentials and affiliations to develop predictions of strategic actions and performance outcomes.

Hambrick and Mason (1984) proposed that executives' experiences, values and personalities affect their:

- field of vision (the directions they look and listen);
- *selective perception* (what they actually see and hear); and
- *interpretation* (how they attach meaning to what they see and hear).

The present study uses UET (Hambrick and Mason, 1984) to provide a comprehensive explanation for the influence of board members' overseas qualifications and work experience on environmental reporting. This is because board members' field of vision, selective perception and interpretation are all likely to be affected by these overseas exposures, especially where it has taken place in culturally distant countries.

Prior studies (Bamber *et al.*, 2010; Ge *et al.*, 2011; Buchholz *et al.*, 2018; Plöckinger *et al.*, 2016; Qi *et al.*, 2018) have widely used UET to explain the impact of CEOs or Chief Financial Officers (CFOs) characteristics on financial reporting. For example, drawing on UET, Bamber *et al.* (2010) found that managers with accounting and finance backgrounds tend to disclose fewer but more precise actual earnings, those with legal backgrounds provided less financial disclosures due to their sensitivity to litigation risk and those born before World War II were more conservative in their disclosures (e.g. more reluctant to forecast). Ge *et al.* (2011) found that accounting choices (e.g. financial reporting) were influenced by CFOs' characteristics; for instance, older CFOs were conservative in accounting choices, whilst those with MBA degrees made less aggressive reporting choices. Expanding the prior literature, our study uses UET to investigate the influences of overseas qualifications and work experience on environmental reporting. Our study, therefore, adds to the understanding of UET by examining the major impact of exposure to the different social, economic and cultural backgrounds through an overseas study on board member attitudes and decision-making towards CED. We, therefore, test the following two hypotheses:

- *H1a.* Companies whose board members earned overseas qualifications have a higher probability of disclosing environmental information than those companies whose board members held only local qualifications.
- *H1b.* Companies whose board members undertook overseas work experience have a higher probability of disclosing environmental information than those companies whose board members did not undertake overseas work experience.

These two hypotheses examine whether companies whose board members earned overseas qualifications (H1a) or overseas work experience (H1b) impact CED. The next three hypotheses test whether different regions where BoDs studied overseas impact CED.

#### 3.2 The impact of overseas study in Anglo and Confucian countries

According to House (2004), Anglo countries share the same ethnicity, language and migration patterns originating centuries ago from regions now identified as Northern Europe (House, 2004). These countries share important characteristics, including English being their national language and being former members of the British Empire, and therefore, heavily influenced by English traditions, such as the use of common law and a parliamentary system of government. Anglo nations also share a high standard of living and quality of life, with high Human Development Index scores (Ashkanasy *et al.*, 2002). Anglo countries also tend to focus on performance and care for the future. People in Anglo countries have less reliance on formal rules and procedures and a greater human orientation,

more equal power distribution and more equality for women than those in other regions (Ashkanasy *et al.*, 2002; House, 2004).

In Anglo countries, high levels of economic development, regulatory pressure and the presence of external board members (outside directors), financial analysts, the press and the media are all key factors in inducing top managers to engage in CSR activities (Pucheta-Martinez *et al.*, 2019). Yu *et al.* (2018) argue that Anglo countries focus on stakeholders, who are active in exerting pressure on companies to operate in an ethical manner. Prior studies, such as Jackson and Apostolakou (2010) and Yu *et al.* (2018), found higher CSR performance levels in Anglo countries than in those countries with coordinated market economies. In addition, education in Anglo countries is likely to promote sustainable development as reflected in the United Nations' Sustainable Development Goals – Target **4.7** (United Nations, 2015).

Expanding the prior literature, our study uses UET to provide a comprehensive explanation for the influences of Anglo cultural traits of board members (in terms of overseas qualifications) on environmental reporting. More specifically, we examine whether these relationships appear to be more pronounced because an overseas study has occurred in Anglo nations, which are culturally distant from a transitional economy such as Vietnam. Our findings, therefore, add to the understanding of UET by demonstrating the major impact of exposure to the different social, economic and cultural backgrounds through an overseas study on board member attitudes and decision-making towards CED. Our study extends prior studies (Post *et al.*, 2011; Pham, 2019a) to investigate the impact of returnees from Anglo countries on the levels of CED in Vietnam, by testing the following two hypotheses:

- *H2a.* Companies whose board members gained qualifications in Anglo Cluster regions have a higher probability of disclosing environmental information than companies with board members who held local qualifications.
- H2b. Companies whose board members gained qualifications in Anglo Cluster regions have a higher probability of disclosing environmental information than companies with board members who held overseas qualifications in other regions.

It is also interesting to explore whether education earned in Confucian regions [8] impacts the environmental reporting of the home country. Confucian countries are settled predominantly by the Chinese (Welch, 2010). Wang and Juslin (2009) argued that due to the influence of Confucianist and Taoist principles, modern Chinese enterprises were supposed to carry on their business harmoniously to become "superior" companies, leading to enhanced CSR activities. However, in practice, various stakeholder groups are dissatisfied with the relatively low level of CED in Confucian countries such as China (Lu and Abeysekera, 2017). Given that there are no prior-related studies that report a significant positive relationship between study in Confucian countries and levels of environmental disclosures, we have stated this hypothesis in the null form:

*H2c.* There is no relationship between the probability of disclosing environmental information and whether board members gained qualifications in Confucian regions or locally.

#### 3.3 Level of education and corporate environmental disclosures

Regarding educational background, it is documented that people who are more educated (i.e. have a higher level of qualifications) should have deeper and broader views and greater

MEDAR levels of moral development than those with less education (Rest and Narváez, 1994). Elm *et al.* (2001) found that graduate students had higher levels of moral reasoning than undergraduates. This ethical perspective should lead to a greater emphasis on sustainability issues including CED. The importance of level of education (e.g. PhD, Masters and Bachelor's degree) has been explored in prior research (Katmon *et al.*, 2017; Kipngetich *et al.*, 2019), although these studies have tended to focus on local qualifications. Expanding on prior literature that examined the level of local study, we have advanced *H1a* to *H3*, to investigate whether the level of overseas degree (*HIE*) affects CED:

*H3.* Companies whose board members obtained overseas-based higher education have a higher probability of disclosing environmental information than companies whose board members do not hold overseas-based higher qualifications.

3.4 Chief executive officer versus chair influences on corporate environmental disclosures Muttakin *et al.* (2018) argued that the influence of the Chair and CEO may be stronger in patriarchal-dominated family companies in emerging countries than elsewhere. Everaert *et al.* (2019) reported that CEOs' ethical ideology and perceived importance of CSR was positively associated with corporate social and environment disclosure (CSD) in Belgian companies. CEO power was found to positively impact CSD in US-listed national commercial banks (Jizi *et al.*, 2014) and in the UK (Li *et al.*, 2018). Moreover, Giannarakis (2014) and Xiao and Yuan (2007) found a significant influence of Chair/CEO duality on CSD. Given the substantial power and influence of CEOs and Chairs in emerging countries (Muttakin *et al.*, 2018; Xiao and Yuan, 2007), the following hypotheses are proposed:

- *H4a.* Companies whose CEOs earned overseas qualifications have a higher probability of disclosing environmental information than those companies whose CEOs did not earn overseas qualifications.
- *H4b.* Companies whose Chairs earned overseas qualifications have a higher probability of disclosing environmental information than those companies whose Chairs did not earn overseas qualifications.

### 4. Research design and methodology

### 4.1 Sample

Our study uses quantitative data collection and analysis. This involves a content analysis of annual, sustainability and integrated reports to capture the quantity of CED and the utilisation of ordered probit models to test our hypotheses. The population of 306 Vietnamese listed firms (VLFs) on the Ho Chi Minh Stock Exchange (HOSE) is investigated, as this had been a trading platform for large-listed firms [9]. The focus on large firms is because of the fact that larger companies are likely to report more information regarding social and environmental activities than others (Cheng *et al.*, 2017; Khalil and O'Sullivan, 2017).

Our study examines the impact of BoDs' overseas study and work experience on CED from 2013 to 2016. In early 2013, the International Financial Corporation (IFC), World Bank Group and State Securities Commission of Vietnam (SSC) jointly issued the first guideline on environmental and social (E&S) reporting for Vietnamese Companies [10], which was based on the GRI framework – G3 (IFC and SSC 2013). This guideline was not binding for Vietnamese firms and was effective from 2013 to 2016. Since December 2016, SSC and IFC

released the second version of guidelines, the E&S Disclosure Guide [11], which was based on G4 and effective from 2017. Companies that were de-listed during the period and listed after the start of 2013 are not included in our study. In addition, companies for which four years' worth of annual reports (2013-2016) could not be located are omitted. This process resulted in 264 companies from 10 sectors [12] being selected for data analysis.

After handling missing data [13], the final number of VLFs examined is 260 across 10 industries in 4 years (2013-2016), yielding a total of 1,040 firm-year observations. We examined 260 company websites, 1,040 annual reports, 60 sustainability reports and 15 integrated reports available across the 4 years (2013-2016). Prior studies have relied on a limited set of disclosure sources, such as company annual reports, sustainability reports (Muttakin *et al.*, 2018; Haque and Deegan, 2010) or have used simple iterative measures of CED such as monetary versus non-monetary disclosures (Rao *et al.*, 2012). Our study is more comprehensive than these prior works as it uses multiple sources of data, such as annual, sustainability and integrated reports and website disclosures and analyses both the quantity and quality of firm CED disclosures. Table 1 shows the breakdown of the sample broken into 10 industries. It indicates that the Industrial and Material sectors represented the biggest proportion of the sample, representing 28.9% and 16.5%, respectively.

### 4.2 Dependent variable: Corporate environmental disclosures

In our study, the level of CED was measured based on a set of 30 environmental items presented in the first sustainable reporting guidelines for VLFs produced by IFC and SSC (2013) and based on the GRI framework (Appendix 3). GRI and its environmental indicators are universally recognised and are widely used in the CED literature (Akrout and Ben Othman, 2016; Kipngetich *et al.*, 2019; Lee, 2017; Marshall and Brown, 2003). For the scoring of CED, four scales were used, with scores ranging from zero to three (Alipour *et al.*, 2019; Meng *et al.*, 2013; Wang *et al.*, 2019; Wiseman, 1982; Zeng *et al.*, 2012). If no disclosure was

Industry	Company <sup>a</sup>	VLFs satisfying criterion for four years observation	% (satisfying criterion for four years observation)	VLFs (after handling missing data)	% (after handling missing data)
Information technology	7	6	2.27	6	2.31
Health care	9	9	3.41	9	3.46
Energy	10	10	3.79	10	3.85
Utilities	18	16	6.06	16	6.15
Financial	19	15	5.68	15	5.77
Consumer discretionary	34	30	11.36	30	11.54
Real estate	35	27	10.23	27	10.38
Consumer staples	35	29	10.98	29	11.15
Materials	52	43	16.29	43	16.54
Industrial	87	79	29.92	75	28.85
Total	306	264	100	260	100

**Notes:** <sup>a</sup>The population of Vietnamese listed firms (VLFs) on the Ho Chi Minh Stock Exchange (HOSE), which is a trading platform for large listed firms. The criteria for large firms are based on Regulation of issuing securities listed at the HOSE (Decision 10/QĐ-SGDHCM dated 13 January 2014 by Ministry of Finance). A company is defined as "large" for a financial year and is listed at the HOSE if it satisfies at least two of the following criteria: the value of equity at the time of listing registration is at least \$6m (US\$); and the minimum of Return on equity (ROE) is 5% for the past two years

Table 1.Sample descriptionclassified accordingto industry

provided on an item, then a score of zero was given. One point was assigned if the disclosure did not include any quantitative information and two points were assigned if the disclosure included quantitative information (numbers or percentages provided or charts or tables given or monetary information). The three-point score was awarded if environmental items were reported in both quantitative and non-quantitative information. Accordingly, a higher score means a higher level of disclosure or more transparency. The maximum achievable score for a company is 90 (30 environmental items, by 3).

Although there have been disagreements about the best ways to measure disclosure levels (Hooks and van Staden, 2011; Lee, 2017; Unerman, 2000), the coding approach applied in our study is consistent with Alipour *et al.* (2019) and Lee (2017), who combined the coding technique by Wiseman (1982) with the GRI framework to evaluate voluntary disclosures in terms of both quantity and quality. As it is necessary to distinguish between poor and better disclosure of each item (Lee, 2017), the coding technique with a score from zero to three for each environmental item allows us to capture a comprehensive picture of CED by evaluating not only how many environmental items were disclosed but also how much detail on each item was reported (Alipour *et al.*, 2019; Lee, 2017).

Our study used each "environmental item" as our unit of analysis. As the presence or absence of particular environmental items was the focus, the "frequency of disclosure" approach was used (Cowen *et al.*, 1987; Haque and Deegan, 2010; Kamal and Deegan, 2013; Ness and Mirza, 1991). To eliminate subjectivity and ensure the reliability of coding, data validation was carried out as follows. Four people were engaged in the score-checking procedure, in which three markers separately calculated CED scores and the fourth reconciled all coding disagreements between the three coders. This process included cross-checking with the scores of the three markers to ensure consistency of CED scores, consistent with Hoang *et al.* (2018) and Saggar and Singh (2017). Where different scores were found for given firms, the scorers discussed the differences until an agreement was reached on an appropriate score. We also computed Cronbach's alpha (0.966) for the three coders, revealing the variance between them to be quite small. The result of Cronbach's alpha suggests a high level of intercoder reliability (Weber, 1990).

It should be noted that to ensure the internal validity of the content analysis, different methods were carried out to ensure the reliability of CED measurement, consistent with Masanet-Llodra (2006) and Passetti *et al.* (2018). The scales of 30 environmental items were also pre-tested using expert opinion [14] (Bachmann *et al.*, 2013).

To improve external validity, we follow the procedure applied by Shima and Fung (2019) to first measure CED by VLFs in one industry in one year, and then extend this to the whole population of VLFs across four years. Moreover, to enhance the generalisation of the research, we follow the approaches of Rivière-Giordano *et al.* (2018) and Shima and Fung (2019) to control for other potential explanatory variables for CED, such as industry, year, firm size or leverage. We also conduct several robustness tests and further analyses to confirm our findings and conclusions (Subsections 4.3 and 4.4).

Although the current GRI framework has developed to be applicable to all industries, there might be some "not applicable" disclosure items for some industries (GRI, 2020). Accordingly, GRI Sector Programme 2020–2022 has been developing industry-specific set of guidelines for 40 sectors; however, to date, GRI G4 Sector Disclosures have been proposed for only 10 prioritised sectors (e.g. Mining and Metals, Oil and Gas and Food Processing) (GRI-Sector, 2020). We observe that to address the issue of not applicable items, some companies discuss their indirect impact of their performance related to some disclosure items, such as EN12, if the disclosure is not directly related to company performance. For example, Bao Viet Holdings, the largest Vietnamese insurance company and Vietnam's

seventh-largest listed company by market capitalisation, disclosed general information related to EN12 items in its sustainability reports (2013–2016). In addition, we expect a different level of disclosure from sensitive industries. It is argued that sensitive industries (e.g. Pulp and Paper, Chemicals, Oil and Gas, Metals and Mining and Utilities) (Patten, 2002; Radhouane *et al.*, 2020) were likely to report more environmental information than less sensitive industries (Cheng *et al.*, 2017). Further, larger companies were more likely to disclose more information related to environmental activities than others (Brammer and Pavelin, 2006; Cheng *et al.*, 2017; Khalil and O'Sullivan, 2017). We, thus, control industry (Cheng *et al.*, 2017; Muttakin *et al.*, 2016) and firm size (Cuadrado-Ballesteros *et al.*, 2015) in our research model.

### 4.3 Independent variables

Personal data of 7,314 board members and CG of 1,040 firm-years were collected from annual reports, HOSE websites and LinkedIn, then summarised on an Excel worksheet. Unlike in developed countries, data of board attributes and CG in Vietnam are not available on the Thomson Reuters DataStream, Osiris Bureau van Dijk and Bloomberg databases. Thus, hand collection of these data was carried out.

The first two independent variables are BMOS and BMOW, measured as the proportion of board members who studied internationally and worked internationally, respectively, over the total board members of the firm, based on Liao et al. (2020) and Ma et al. (2019). Other independent variables are the proportion of board members with *HIE*, overseas MBA (BMBA)/law (BLAW)/accounting (BACC) degrees, international qualifications in Anglo Cluster regions (BSA); and overseas degrees in Confucian Asian regions (BSC). We used this measurement to ensure the consistency with the control variables in our model, such as the proportion of independent directors on a board (Cui et al., 2020; Liao et al., 2015; Rao et al., 2012) and woman directors on a board (Bear, 2010) and the percentage of board members with multiple directorships over the total board members of the firm (Rupley *et al.*, 2012; Sun et al., 2019). However, BMOS or BMOW can be computed as continuous variables based on the number of years that board members studied or worked overseas (Carpenter et al., 2001; Sambharya, 1996; Sullivan, 1994), as the influence of international experience could be greater where the exposure was for a longer period of time. Thus, we also used years of overseas study/work experience in our robustness check as different measurements of independent variables to confirm the impact of BMOS/BMOW on CED. Separate from BMOS/BMOW, CHAS and CEOS were measured as dichotomous variables as a firm has only one Chair and one CEO but many board members.

Our study used 10 regional cultural clusters, based on House (2004) and cited by Karaibrahimoglu and Cangarli (2016, p. 73), to classify regions where board members studied. These cultural clusters, as defined by House (2004), are based on cultural values, country economic development and linguistic nations across administrative borders. As such, this cultural cluster approach addresses some of the main criticisms to Hofstede's framework. Hofstede's cultural dimensions were criticised by previous studies such as Baskerville (2003), Khlif (2016), as they ignored multicultural countries, for example, Anglo (Australia, Canada, England, Ireland, New Zealand and the USA) or Confucian regions (China, Hong Kong and South Korea). Thus, Hofstede's model is likely to be inapplicable to all countries (Khlif, 2016; López-Duarte *et al.*, 2016). In particular, in transitional countries such as Vietnam where society has witnessed significant changes compared to 30 years ago (Orij, 2010), Hofstede's cultural model is not useful in explaining accounting phenomena, and thus, shows methodological weaknesses (Baskerville, 2003).

### MEDAR 4.4 Research model

A regression model was run where *CED* was the dependent variable and *BMOS*, *BMOW*, *HIE*, *BMBA*, *BLAW*, *BACC*, *BSA*, *BSC*, *CEOS* and *CHAS* were independent variables. Three types of control variables were used in the study, namely, board characteristics (Ismail and Latiff, 2019; Ma *et al.*, 2019; Rao and Tilt, 2016), ownership types (Hoang *et al.*, 2018; Xiao and Yuan, 2007) and firm features (Cheng *et al.*, 2017; Muttakin *et al.*, 2018), which are defined in our research model. Specifically, *SO*, *FO*, *BSO*, *MDIR*, *BIND*, *BAGE*, *BGEN*, *EPS*, *EPW* and firm features (i.e. *FSIZE*, *FAGE*, *IND*, *YEA*, *LEV*) were the control variables:

$$\begin{split} CEDij &= \beta_0 + \beta_1 BMOSij + \beta_2 BMOWij + \beta_3 HIEij + \beta_4 BMBAij + \beta_5 BLAWij \\ &+ \beta_6 BACCij + \beta_7 BSAij + \beta_8 BSCij + \beta_9 CHASij + \beta_{10} CEOSij \\ &+ \beta_{11} MDIRi + \beta_{12} SOij + \beta_{13} FOij + \beta_{14} BSOij + \beta_{15} BINDij \\ &+ \beta_{16} BAGEij + \beta_{17} BGENij + \beta_{18} EPSij + \beta_{19} EPWij + \beta_{20} FSIZEij \\ &+ \beta_{21} FAGEij + \beta_{22} INDij + \beta_{23} YEARj + \beta_{24} LEVij + \varepsilon_{ij} \end{split}$$

where CEDi is the total score of CED for firm *i* (*i* is the Vietnamese listed company and *i* is the year of observation (i = 1, 2, 3, 4); BMOS is the proportion of board members with overseas qualifications over the total board members of the firm; BMOW is the proportion of board members with overseas work experience over the total board members of the firm (Herrmann and Datta, 2005; Zhuang et al., 2018); HIE is the proportion of board members with overseas higher education over the total board members of the firm: BMBA or BLAW are the proportion of board members with overseas MBA or law degrees over the total board members of the firm; BSA is the proportion of board members with international qualifications in Anglo Cluster regions over the total board members of the firm; BSC is the proportion of board members with international qualifications in Confucian Asia regions over the total board members of the firm: CHAS = 1 if Chair studied overseas and 0 otherwise; and CEOS = 1 if CEO studied overseas and 0 otherwise. SO is the percentage of shareholding owned by the State over the total issued shares of the company (Hoang et al., 2018); FO is foreign ownership measured by the percentage of shareholding held by foreign investors (Xiao and Yuan, 2007); BSO is the percentage of shares owned by each board member over the total issued shares (the same measurement has been used by Eng and Mak (2003), Ghazali (2007) and Khan et al. (2013)); BIND is the proportion of independent directors on a board, which is also consistent with many prior studies (Cui et al., 2020; Liao et al., 2015; Rao et al., 2012); BAGE is mean board age (Ismail and Latiff, 2019; Ma et al., 2019); BGEN is the proportion of female directors over the total directors of the firm (Rao and Tilt, 2016); MDIR is the proportion of board members with multiple directorships over the total board members of the firm (Rupley et al., 2012; Sun et al., 2019); FSIZE is a logarithm of firm's sales; FAGE is the natural log of the number of the year, which firms listed; IND is industry; YEA is the year of observation; LEV is leverage of the listed firm (Cheng et al., 2017; Muttakin et al., 2018); and *\varepsilon ij* is an error term.

The environmental performance index (EPI) of countries where boards studied (*EPS*) and of countries where boards worked (*EPW*) are also control variables. The reason for controlling these variables is that corporate environmental performance has been found to be associated with countries' commitment to environmental policy and law (Clarkson *et al.*, 2008; De Villiers and Marques, 2016; Iatridis, 2013). We treated *EPS* and *EPW* as two

separate variables, as countries where board members studied overseas were sometimes different from those where board members worked overseas. *EPS* and *EPW* were computed by the average of the EPI of countries where boards studied and worked, respectively. EPI has covered more than 180 countries and is released every second year (CIESIN, 2018). For the period of study (2013–2016), EPI 2014 and EPI 2016 were available on the database by Hsu (2016) and Yale University (2014). EPI was created by The Yale Centre for Environmental Law and Policy and the Centre for International Earth Science Information Network at Columbia University (CIESIN, 2018). The index can range from 0 to 100; with higher values indicating countries that strongly engage in environmental schemes (CIESIN, 2018).

Hypotheses were tested by running panel data to determine whether overseas study and work experiences earned by board members could impact on CED in the period from 2013 to 2016. The models were tested using STATA Release 15 with different estimation techniques (Cameron and Trivedi, 2009).

### 5. Results and discussion

#### 5.1 Descriptive statistics

Based on the results of pairwise correlation statistics, reported in Appendix 1, it is concluded that multicollinearity did not appear to influence the empirical models in our research (Darnall *et al.*, 2010; Hair *et al.*, 2011). A summary of descriptive statistics for CED, board and firm characteristics in Vietnam for the period of 2013–2016 are reported in Table 2. Results in Table 2 reveal that the average CED level was low in Vietnam, as CED scores ranged from 0 to 73, with a mean of only 3.65.

The finding of low CED in Vietnam is consistent with multinational studies by Pucheta-Martínez *et al.* (2019) and Cui *et al.* (2020), which documented companies located in emerging nations possess lower levels of CED than those in liberal and developed countries. However, the level of CED in Vietnam appears to be even lower than that found in other emerging countries, such as China [15], Iran [16] and India [17] (Alipour *et al.*, 2019; Cheng *et al.*, 2017; Garg and Kumar, 2018).

Table 2 shows that *BMOS* had a mean of 15%, whilst *BMOW* had a mean of 8%. This indicates for VLF, 15% of their board, on average, studied overseas and an even lower percentage of those undertook international work experience (8%). This reveals that a minority of board members have had the opportunity to study or work overseas. We calculated *BMOS* and *BMOW* for the top 10 most transparent VLFs as voted by Forbes (2016) and compared them to those of the whole population. Results in Table 2 indicate that amongst this top 10, on average, 63% of board members studied overseas, whilst 39% undertook international work experience. This is much higher than for the rest of the sample where, on average, 15% of BoDs studied overseas and 8% worked overseas.

Closer inspection of *BMOS* and *BMOW* reveals that the highest proportion of directors studied and worked overseas in Anglo Cluster regions (about 43%), with the US being the leading country, followed by Australia and England (Appendix 2). This descriptive finding reflects the historical circumstances of Vietnam. Sponsorship programmes of Australian, US and European Governments targeted to developing countries have resulted in considerable growth in international education amongst Vietnamese people (Auletta, 2000; Rosenau, 2007). These initiatives have enabled many Vietnamese citizens to study in the West rather than in other regions. Our results in Table 2 also indicate that, on average, the number of board members who obtained qualifications in Anglo Cluster regions (5%) was five times than those who gained degrees in Confucian Asia (1%). Interestingly, the percentage of Chairs or CEOs studying overseas was only 1%, indicating that VLFs were more likely to possess Chairs or CEOs with local qualifications. The impacts of these variables are tested and reported in the next section.

MEDAR
-------

Variable	Obs	Mean	SD	Minimum	Maximum
Dependent variab	ole				
CED	1,040	3.65	8.09	0.00	73.00
Independent varia	ables				
BMOS (%)	1,040	15.00*	23.00	0.00	100
BMOW (%)	1,040	8.00*	13.00	0.00	63
BSA (%)	1,040	5.00	12.00	0.00	100
BSC (%)	1,040	1.00	10.00	0.00	100
HIE (%)	1,040	8.00	15.00	0.00	100
BMBA (%)	1,040	4.00	10.00	0.00	80
BACC (%)	1,040	1.00	4.00	0.00	29
BLAW (%)	1,040	1.00	3.00	0.00	25
CHAS**	1,040	1.00	11.00	0.00	100
CEOS	1,040	1.00	12.00	0.00	100
Type of directors			Chair		(%)
Independence			88		8.46
EXE			685		65.87
NON-EXE			267		25.67
Total			1,040		100
Share ownership	(SHO) (%)				
0			233		22.40
>0			807		77.60
>10			235		22.60
>30			49		4.72
>50 <sup>a</sup>			13		1.25
Duality					
Separate			747		71.83
Chair = CEO			293		28.17
Total			1,040		100

Notes: Dependent Variable. CED is the total score of CED for the firm; where CED was categorised into four groups as follows: Group 1 (50% of the total sample) with a score of 0, called "non-disclosures"; Group 2 with values from 1-4 (25% of the total sample), named "minimum disclosures"; Group 3 with scores from 5–11 (15% of the total sample), called "medium disclosures"; and Group 4 with scores above 11 (10% of the total sample), called "high disclosures". Independent Variables. BMOS is the proportion of board members with overseas qualifications over the total board members of the firm; BMOW is the proportion of board members with overseas work experience over the total board members of the firm; HIE is the proportion of board members with overseas higher education over the total board members of the firm; BMBA/BLAW/ BACC is the proportion of board members with overseas MBA/Law/Accounting degrees over the total board members of the firm; BSA is the proportion of board members with international qualifications in Anglo Cluster regions over the total board members of the firm; BSC is the proportion of board members with international qualifications in Confucian Asia regions over the total board members of the firm; CHAS = 1 if Chairman studied overseas and 0 otherwise; and CEOS = 1 if Chief executive officer (CEO) studied overseas and 0 otherwise. \*BMOS and BMOW in the high disclosure group (10% of the highest CED scores in the population) were 27% and 10%, respectively. In the top 10 most transparent VLFs as voted by Forbes (2016), BMOS and BMOW were 63% and 39%, respectively. \*\*Further analysis of Chairs in VLFs. <sup>a</sup>The highest percentage was 64.74%

Table 2.Descriptive statistics

### 5.2 Order probit results

The dependent variable, CED, was heavily, positively skewed with many scores of zero. Although many forms of transformations of this variable were attempted (Becker *et al.*, 2018; Osborne, 2010), all results indicated that the dependent variable was heavily positively skewed, and therefore, non-normally distributed. We, therefore, decided to use the approach

of Clark *et al.* (2001) to group our dependent variable into four proportional groups. Given the ordered nature of our dependent variable, we decided to run an ordered probit estimation consistent with studies such as Gray *et al.* (2006) and Srivastava *et al.* (2017). The use of such a model and grouping of the dependent variable were consistent with prior studies of corporate disclosures where researchers faced similar issues of extreme skewness (Acquisti *et al.*, 2012; Kim and Lyon, 2011).

Corporate environmental disclosures

Results of the four ordered probit models are reported in Table 3. All models were significant (Prob > Chi(2) = 0.000). Model 4 was finally chosen as it showed the highest log-likelihood of 411.78 and Pseudo  $R^2$  of 25.29% [18] (Table 3). Next, Table 4 reports the results of the original ordered probit Model 4 with marginal effects. The first column of Table 4 repeats the results of Model 4, which are presented in Table 3. The next columns of Table 4 (e.g. from Columns 2 to 5) report the results of marginal effects in which coefficients were computed and analysed whilst the outcomes changed across four different categories of CED from the "non-disclosures" group to the "high disclosures" group. Results of marginal effects exhibit positive coefficients for *BMOS*, *BSA* and *BMBA*, which means an increase in

Variables	Expected signs	Model 1 (add BMOS and BMOW)	Model 2 (add HIE, BMBA, BLAW)	Model 3 (add regions of study)	Model 4 (add CHAS, CEOS)
CED					
BMOS	+	$0.33^{**}(0.14)$	0.28 ** (0.14)	1.07*** (0.33)	$0.84^{**}(0.41)$
BMOW	+	-0.02(0.11)	-0.01(0.13)	0.04 (0.11)	0.01 (0.02)
HIE	+	(,	0.05 (0.06)	-0.02(0.04)	-0.02(0.04)
BMBA	+		2.06* (0.12)	2.03* (0.03)	3.03* (0.05)
BLAW	_		$-6.62^{**}(0.05)$	$-7.78^{**}(0.60)$	$-8.79^{**}(0.59)$
BACC	+		-1.40(0.29)	-1.01(0.51)	-1.47(0.36)
BSA	+			0.37*** (0.16)	1.48*** (0.66)
BSC	-			-0.33(0.27)	-0.28(0.88)
CHAS	+				0.57*** (0.19)
CEOS	+				-2.46(0.38)
Control variables					
SO	$\pm$	-0.001(0.00)	0.01 (0.01)	0.01 (0.01)	0.01 (0.00)
FO	<u>+</u>	$0.02^{**}(0.01)$	$0.03^{***}(0.01)$	$0.03^{***}(0.01)$	$0.03^{***}$ (0.01)
BAGE	+	-0.001(0.00)	0.23 (0.00)	0.01 (0.00)	0.01 (0.00)
BSO	<u>+</u>	$-0.01^{**}(0.01)$	$-0.01^{**}(0.01)$	-0.01*(0.01)	$-0.11^{***}(0.01)$
BGEN	<u>+</u>	0.005 (0.10)	0.17 (0.26)	0.05 (0.27)	0.02 (0.30)
BIND	+	-0.02(0.03)	-0.01(0.03)	-0.01(0.03)	-0.03(0.03)
MDIR	+	0.05* (0.10)	0.47* (0.06)	0.58* (0.07)	0.66* (0.14)
EPS	+	-0.02(0.05)	-0.04(0.01)	-0.03(0.06)	-0.05(0.07)
EPW	+	-0.03(0.04)	-0.07(0.03)	-0.05(0.02)	-0.06(0.05)
FSIZE	+	0.15*** (0.05)	$0.18^{***}(0.05)$	0.14** (0.06)	0.12** (0.06)
FAGE	+	0.18 (0.15)	0.19 (0.18)	0.14 (0.18)	0.03 (0.18)
LEV	_	$-0.01^{***}(0.04)$	$-1.26^{***}(0.33)$	$-1.34^{***}(0.32)$	$-1.33^{***}(0.32)$
Industry		Yes	Yes	Yes	Yes
Year dummy		Yes	Yes	Yes	Yes
Observations		1,040	1,040	1,040	1,040
Log likelihood		-604.57	-445.13	-439.7	-411.78
Pseudo $R^2$		16.24%	17.48%	20.66%	25.29%

Table 3.

**Notes:** \*, \*\* and \*\*\*denote significance at the 0.10, 0.05 and 0.01 levels, respectively. Standard errors in parentheses

Results of ordered probit models

### 

MEDAR	Variables	Expected signs	(1) CED	(2) Pr(Non- disclosure)	(3) Pr(Minimum disclosure)	(4) Pr(Medium disclosure)	(5) Pr(High disclosure)
Table 4. Results of ordered probit Model 4 with marginal effects	BMOS BMOW BSA BSC HIE BMBA BLAW BACC CHAS CEOS Notes: St respective	+ + - + + + + + + y	$\begin{array}{c} 0.84^{**} \ (0.41) \\ 0.01 \ (0.02) \\ 1.48^{***} \ (0.66) \\ -0.28 \ (0.88) \\ -0.02 \ (0.04) \\ 2.06^{*} \ (0.12) \\ -8.79^{**} \ (0.59) \\ -1.47 \ (0.36) \\ 0.57^{***} \ (0.19) \\ -2.46 \ (0.38) \\ \end{array}$	-0.192* (0.11) 0.12 (0.03) -0.517*** (0.19) 0.007 (0.26) 0.004 (0.01) -0.64* (0.34) 2.74*** (0.78) 0.48 (0.44) -0.18*** (0.06) 0.79 (0.36) es. *, ** and ****de	$\begin{array}{c} 0.040^{*} (0.02)\\ 0.14 (0.05)\\ 0.109^{**} (0.04)\\ -0.001 (0.05)\\ -0.001 (0.00)\\ 0.14^{*} (0.08)\\ -0.61^{**} (0.18)\\ -0.12 (0.11)\\ 0.036^{**} (0.01)\\ -0.21 (0.35)\\ \end{array}$	$\begin{array}{c} 0.072^* \ (0.04) \\ 0.07 \ (0.12) \\ 0.195^{**} \ (0.08) \\ -0.003 \ (0.10) \\ -0.002 \ (0.00) \\ 0.22^* \ (0.11) \\ -0.95^{**} \ (0.29) \\ -0.17 \ (0.15) \\ 0.067^{***} \ (0.02) \\ -0.28 \ (0.35) \\ e \ at \ the \ 0.10, \ 0.0 \end{array}$	$\begin{array}{c} 0.079^{*} \left( 0.04 \right) \\ 0.18 \left( 0.01 \right) \\ 0.213^{***} \left( 0.08 \right) \\ -0.003 \left( 0.11 \right) \\ -0.002 \left( 0.01 \right) \\ 0.28^{*} \left( 0.16 \right) \\ -1.18^{**} \left( 0.37 \right) \\ -0.18 \left( 0.17 \right) \\ 0.077^{***} \left( 0.03 \right) \\ -0.3 \left( -0.34 \right) \\ 5 \text{ and } 0.01 \text{ levels,} \end{array}$

BMOS, BSA and BMBA, certainly decrease the probability of being in the lowest category (i.e. non-disclosure group) and increase the probability of being in the highest category (i.e. high disclosure group) (Cameron and Trivedi, 2009, p. 477).

5.2.1 The impact of overseas study and work experience on corporate environmental *disclosures.* Results in Tables 3 and 4 support *H1a*, which states that the probability of companies disclosing environmental information whose board members earned overseas qualifications was higher than for those without overseas qualifications. This finding highlights that exposure to overseas qualifications makes these individuals more aware of the importance of improving CED. Giannetti et al. (2012) suggested that directors with overseas education are likely to be more focussed on improving their CG through higher disclosure levels; in turn, better-governed firms are likely to be more transparent for their stakeholders.

However, the results in Tables 3 and 4 did not support H1b in that the overseas work experience of board members did not significantly impact on CED in Vietnam. This finding is inconsistent with prior studies, for example, Carpenter et al. (2001) and Zhuang et al. (2018), that reported a positive relationship between overseas work experience and sustainability performance. This finding also implies that the effect of overseas education appears to be more important in influencing levels of CED in Vietnam than does overseas work experience. This unexpected result reflects the historical circumstances of Vietnam. Vietnam's economic reforms since 1986 and a colonial past (e.g. 20 years under the control of the USA) have led to a significant increase in the numbers of Vietnamese studying overseas (Oliver and Nguyen, 2010). Further, all sponsorship programmes [19] targeted to developing countries (e.g. Vietnam) require awardees to return to their home country and use their overseas skills to contribute to Vietnam's development (Auletta, 2000; Rosenau, 2007). This requirement has enabled many Vietnamese to study overseas rather than undertake overseas work experience, leading to a low number of board members working overseas (6.2%) of the board population). In addition, this small proportion of board members undertaking international work experience may result in "brain drain" in Vietnam, whereby Vietnamese nationals who secure employment overseas are unlikely to return to work in Vietnam (Docquier and Rapoport, 2012; Nguyen, 2014).

5.2.2 The impact of overseas study in Anglo and Confucian countries. Given that the same measurement approach and scale were applied for two variables (e.g. BSA and BMOS) in all four disclosure groups of CED, the coefficients of BSA (51.7%; 10.9%; 19.5%; 21.3%) were higher than those of BMOS (19.2%; 4%; 7.2%; 7.9%). The interpretation of these results means that, for example, firms whose board members studied in Anglo Cluster regions were 21.3% more likely to be a member of the high disclosure group than firms whose board members studied locally. Meanwhile, the probability that a firm would be a member of the high disclosure group when its board members studied overseas in other regions was only 7.9% more likely than a firm whose board members studied locally. Thus, firms with BoDs studying in Anglo Cluster regions were more likely to report higher levels of environmental information than those with board members studying overseas in other regions. Results from Tables 3 and 4 support H2a and H2b, indicating that the probability of companies disclosing environmental information whose board members gained qualifications in Anglo Cluster regions was higher than for those with board members studying locally and studying overseas in other regions. These findings are consistent with prior studies (Cui et al., 2020; Jackson and Apostolakou, 2010; Pucheta-Martínez et al., 2019), which found higher social and environmental commitments in Anglo Cluster nations. However, unlike these previous studies, our study contributes to the literature by showing a significantly positive link between BSA and environmental reporting in the transitional country of Vietnam. This result suggests that managers' favourable attitudes towards environmental issues are also influenced by their cultural background (Bechtel *et al.*, 1999; Schwartz and Bardi, 2001). Our study highlights the value of Anglo education in terms of enhancing CED in transitional countries, where different attitudes towards environmental issues seem to exist (Pham, 2019a; VanDonkelaar, 2013).

Results in Tables 3 and 4 indicate that we are not able to statistically reject the null Hypothesis *H2c*. More specifically, there is not enough evidence to support that possession of board members who studied in the Confucian Asia regions would significantly influence the CED level in Vietnam. This finding builds on prior studies, for example, Lu and Abeysekera (2017) and Situ and Tilt (2018), which found low sustainability disclosures in China. In addition, this expected outcome does not agree with Wang and Juslin (2009), who argued that the moral standards of traditional Chinese wisdom (e.g. the influence of Confucianism and Taoism principles) could enhance CSR activities. One possible justification for this result is that Confucian countries (e.g. China and Hong Kong) are settled predominantly by the Chinese, who are likely to possess the same cultural values as Vietnamese populations (Welch, 2010). Another possible explanation for this finding is that Vietnam and China possess similar educational systems, which makes returnees from China unlikely to change their views of CED. A notable characteristic of Vietnam's education system was a long history of Chinese Confucian influences, which commenced when the Chinese first established their control in Vietnam and continued to develop as an official ideology under the Vietnamese Kingdoms (Shao-hui, 2009). In addition, Vietnam shares similar political (e.g. Communist party) and economic systems (e.g. transitional economy) to China (Elliott, 2016). Vietnam and China used to perform "a traditional model of socialist accounting", which came from the Soviet Union (Adams and Do, 2003, p. 44). This finding of BSC implies that socialist accounting systems are less likely to consider sustainability reporting as critical.

5.2.3 Level of education and corporate environmental disclosures. A finding worthy of attention is that the level of overseas degrees (e.g. PhD, Master's and Bachelor's degree) did not significantly impact the levels of CED in Vietnam. This result did not support H3 as no statistically significant relationship between HIE and CED was found (Tables 3 and 4). This

outcome is consistent with Post *et al.* (2011), who also found that firms in the USA with directors with higher degrees were unlikely to show higher levels of CSR. Our finding of an insignificant effect of *HIE* on CED is inconsistent with prior studies in emerging countries, such as Katmon et al. (2017) and Kipngetich et al. (2019), which found a significantly positive relationship between board members' education levels and CED in Malaysia and CED in Kenva, respectively. However, it is worth noting that these two studies examined local education rather than overseas degrees obtained by board members. The inconsistent result for *HIE* in Vietnam reflects a different context because of the distinct characteristics of directors' overseas education. It should be noted that board members with overseas bachelor's degrees tend to stay overseas longer (e.g. four years) than those with international master's degrees (one or two years). VLFs' BoDs primarily held overseas master's degrees (49.57%) whilst 36.77% of directors earned international bachelor's degrees (Appendix 2). In addition, findings related to overseas study in our research might be different from findings related to local education by Katmon et al. (2017) and Kipngetich et al. (2019). The effects of overseas higher education on CED may depend on how long BoDs lived overseas in terms of their ability to gain a broader perspective and different patterns of thinking or be affected by the culture of the country in which they studied (Carpenter et al., 2001; Sambharya, 1996).

5.2.4 Chief executive officer versus chair influences on corporate environmental disclosures. Results in Tables 3 and 4 reveal that H4a was not statistically supported; thus, having a CEO who gained international qualifications insignificantly influenced CED. This unexpected finding is contrary to prior studies (Muttakin *et al.*, 2018; Everaert *et al.*, 2019; Li *et al.*, 2018), which reported CEOs' impacts on CED levels. However, results in Tables 3 and 4 support H4b, and indicate that having a Chair who studied overseas significantly impacted CED in Vietnam. This finding implies that Chairs in Vietnam tend to have more influence on sustainability reporting than in other Asian countries, such as Bangladesh (Muttakin *et al.*, 2018), where CEO power was found to be dominant.

To examine the power of Chairs in Vietnam, further analysis of share ownership in Table 2 demonstrates that 77.6% of the Chairs owned shares in their listed firms, with 22.6% of them holding more than 10% of shares and 5% of them owning more than 30%. Some Chairs held as much as 64.74% of their companies' shares. These results suggest that Chairs in Vietnam have high percentages of ownership in their listed companies; thus, they are likely to seek to exercise more influence over decision-making. Moreover, regarding the types of directors shown in Table 2, nearly 92% of Chairs were not independent (66% of them were executives and 26% were non-executive directors). Amongst the 66% of Chairs who were executive directors, 28% were CEOs; the rest were managers or were involved in management. The high duality in VLFs is different from good CG practice in developed countries [20], which highlights that a Chair should not also be a CEO of the listed company. In contrast, the *Vietnamese Law on Enterprises 2014* [21] permits this practice.

CEO duality was then added to test its impact on CED, whilst *BSO* was one of the control variables in the model (Barako *et al.*, 2006; Xiao and Yuan, 2007). The result for CEO duality was statistically insignificant, whilst *BSO* negatively influenced CED. The outcome of duality in Vietnam is inconsistent with previous research in the USA by Giannarakis (2014) and in China by Xiao and Yuan (2007), which showed a negative influence of CEO duality on CSD. However, this finding of duality in Vietnam is consistent with prior studies in Hong Kong by Ho and Wong (2001) and in Kenya by Barako *et al.* (2006), which documented that duality did not statistically or significantly influence voluntary disclosures. A possible reason for the insignificant effect of CEO duality on CED in Vietnam is that a Chair who served as CEO of a listed firm was often also a substantial shareholder, as previously discussed. The same situation was found for VLFs, where 77.6% of the Chairs were

shareholders and many held substantial numbers of shares. Therefore, it does not matter whether the two positions are separated (Ho and Wong, 2001). Further, the result of a negative effect of *BSO* on CED emphasises that the power of the Chairs in Vietnam is more likely to stem from their high ownership in VLFs, which made them less independent.

5.2.5 Additional analysis. We further analysed the influence of different fields of overseas study on CED. The results presented in Tables 3 and 4 indicate that the proportion of board members with overseas MBA degrees significantly positively impacted CED. In contrast, the percentage of directors with overseas legal degrees negatively influenced CED levels. These results related to overseas study in the fields of MBA and law are consistent with Lewis *et al.* (2014); however, this prior study investigated the impacts of local education on CED. BoDs with MBA degrees are trained to make strategic decisions and be aware of the need for improved CSR (Bertrand and Schoar, 2003). Meanwhile, top managers with legal qualifications are concerned about the potential costs of CED and are, therefore, reluctant to voluntarily report environmental information (Lewis *et al.*, 2014). Further, given that accounting programmes include the topics of social and environmental reporting (Deegan, 2012), accounting students are more likely to understand the value of CED and GRI guidelines than other students. However, our results in Tables 3 and 4 reveal an insignificant impact of accounting studies on CED.

5.2.6 Testing for the endogeneity problem and robustness checks. An endogeneity problem might occur in the relationship between board diversity and CED because of omitted variable bias or reverse causality (Ben-Amar et al., 2017; Katmon et al., 2017). A Hausman test including two stages was performed to examine whether CED and BMOS/ BMOW were endogenous (Gul and Leung, 2004; Sundaramurthy et al., 2014). Firstly, the regression of BMOS/BMOW [22] on all exogenous variables and an instrument variable (IV) was performed to obtain a predicted residual. The IV was the international integration year in Vietnam (i.e. 1995 when the formal normalisation of US-Vietnam diplomatic relations took place). The selection criteria for this IV was based on Zhang et al. (2018). Udalov et al. (2017) and Katmon et al. (2017), as this variable highly correlated with BMOS/BMOW but not with CED. An additional check confirmed whether the international integration year is a good IV by comparing  $R^2$  values and F-statistics in the first-stage regression without IV and with IV (Papies et al., 2017; Ullah et al., 2020). Results in Table 5 indicate that  $R^2$  values and Fstatistics in the first-stage regression with IV increased significantly. In particular, Fstatistics increased 38% compared to the model without IV (from 359.33 to 494.91). Another reason for choosing this IV is because formal normalisation of US-Vietnam exogenously changed the supply of potential directors with overseas experience for VLFs. A change in

	Without IV	With IV	
First stage: Checking for a good instrumental R <sup>2</sup> F-statistics	variable (IV) 90.11% 359.33	90.63% 494.91	
Coet. Second stage: Results of the residuals	0.05*** BMOS	BMOW	Table 5.   Testing for the   endogeneity problem –
Coefficient Number of observations	0.0586 1,040	-0.056 1,040	results of the Hausman test and Durbin Wu-Hausman
Note: ***denote significance at the 0.01 leve	1		test

US-Vietnam diplomatic relations marked a significant rise in studying and working overseas when Vietnam achieved substantial economic growth through international integration (Oliver and Nguyen, 2010). This IV is a dummy variable that equals one if a board member studied/worked overseas after 1995 and equals zero otherwise. Our sample indicates that most board members studied/worked overseas after 1995.

Secondly, the residual of *BMOS/BMOW* retrieved from this first regression was used as an additional regressor in the next model where CED was the dependent variable, whilst *BMOS/BMOW* and the residual of *BMOS/BMOW* were independent variables; other board characteristics (i.e. *BAGE, BIND, BGEN and BSO*) were control variables. In this second stage, the model was estimated using the user-written Conditional Mixed Process Command in Stata, which was developed by Roodman (2018) for an ordered probit model. Results in Table 5 show that the residuals of *BMOS/BMOW* were statistically insignificant (*p*-values were 0.515 and 0.505, respectively) whilst the coefficient of residuals *BMOS* and *BMOW* were not significantly different from zero (0.0586 and -0.056, respectively). Based on Papies *et al.* (2017) and Ullah *et al.* (2020), these findings show that an endogeneity issue did not exist in our model. Further, we ran the Durbin Wu-Hausman tests for endogeneity by using STATA command "estat endogenous" based on Ullah *et al.* (2020) and the same results were found.

Various robustness checks [23] focussing on estimating the effects of CED were undertaken. Results of robustness tests in Table 6 are consistent with our preliminary results presented in Tables 3 and 4. Moreover, we run a nonparametric test (i.e. Mann-Whitney U Test) to further investigate whether the group of BoDs with overseas work experience had a lower or higher score of CED than those with the overseas study. The results of Mann-Whitney U Test in Table 6 indicate that the mean CED for the two groups were significantly different (*p*-value = 0.000 < 0.05) with it being much higher for the overseas study group. Further, the group of board members with overseas work experience had a significantly lower mean score for CED (0.90) than those with overseas study (1.06). Thus, the effect of overseas education appears to be more important in influencing levels of CED in Vietnam than does overseas work experience. Another Mann-Whitney U test was performed as a robust test for the impacts of CEO and Chair on CED (Table 6). Given an absence of endogeneity in our model and results of robustness tests, the ordered probit estimation results in Table 3 and marginal effects in Table 4 satisfactorily explain the impacts of overseas study on CED.

We also ran an additional robustness test using Hofstede's model. Instead of using 10 regional cultural clusters, based on House (2004), we used two of Hofstede's cultural dimensions, namely, individualism and long-term orientation, as they are linked to sustainability behaviour, according to Orij (2010) and Khlif *et al.* (2015). Results of Model 6 (long-term orientation) and Model 7 (individualism) in Table 6 reveal that the Pseudo  $R^2$  of Models 6 and 7 (13.3%) that used Hofstede's cultural dimensions were much lower than those of the first five models (from 18.25% to 21.55%), which used regional cultural clusters, based on House (2004). Given that Pseudo  $R^2$  measures the goodness of fit, the higher Pseudo  $R^2$  indicates better fit and predictability (Ben-Amar *et al.*, 2017; Hagle and Mitchell, 1992). The results of *BMOS/BMOW* and Hofstede's cultural dimensions in Models 6 and 7 were different from those found in the first five models. This indicates that House (2004) is a better fit for our study than Hofstede.

### 6. Conclusion

Our study empirically examined the influence of boards' overseas qualifications and work experience on CED in state-led transition economies, such as Vietnam. Using ordered probit models, we found that exposure to overseas perspectives, especially from the Anglo nations,

Variables	Model 1 (change variables)	Model 2 (change variables)	Model 3 (change variables)	Model 4 (change measurements)	Model 5 (change measurements)	Model 6 Hofstede's model	Model 7 Hofstede's model
BMOS BMOW BSA BSC LTO	$\begin{array}{c} 0.401^{*} \left( 0.24 \right) \\ 0.008 \left( 0.05 \right) \\ 1.404^{****} \left( 0.68 \right) \\ -0.0007 \left( 0.14 \right) \end{array}$	$\begin{array}{c} 0.405^{*} \left( 0.49 \right) \\ 0.044 \left( 0.25 \right) \\ 0.238^{*} \left( 0.22 \right) \\ 0.035 \left( 0.20 \right) \end{array}$	$0.39^{**} (0.15)$ $0.238^{*} (0.27)$	$\begin{array}{c} 0.604^{*} \ (0.49) \\ 0.038 \ (0.25) \\ 0.0038^{***} \ (0.22) \\ -0.12 \ (0.20) \end{array}$	$\begin{array}{c} 0.17^{*} \left( 0.24 \right) \\ 0.303 \left( 0.07 \right) \\ 0.005^{***} \left( 0.67 \right) \\ 0.303 \left( 0.07 \right) \end{array}$	0.48 (0.39) 0.76 (0.5) 0.01 (0.01)	0.48 (0.39) 0.78 (0.5)
1DV HIE BMBA BLAW BACC CHAS Pseudo R <sup>2</sup>	$\begin{array}{c} -0.007 (0.86)\\ 1.08^{*} (0.13)\\ -3.56^{**} (0.04)\\ -1.43 (0.25)\\ 0.48^{***} (0.18)\\ 18.25\% \end{array}$	$\begin{array}{c} 0.044 \ (0.06) \\ 0.64^{*} \ (0.07) \\ -2.76^{**} \ (0.06) \\ -1.10 \ (0.32) \\ 0.48^{**} \ (0.16) \\ 20.94\% \end{array}$	$\begin{array}{c} 0.009 & (0.06) \\ 0.86,^{*} & (0.18) \\ -1.98^{**} & (0.14) \\ -1.53 & (0.26) \\ 0.40^{***} & (0.17) \\ 19.01\% \end{array}$	$\begin{array}{c} 0.33 \\ 0.42^{**} \\ 0.42^{**} \\ 0.18 \\ 21.55\% \end{array}$	$\begin{array}{c} -0.019 \ (0.04) \\ 1.32^{*} \ (0.15) \\ -4.21^{**} \ (0.08) \\ 1.08 \ (0.97) \\ 0.37^{**} \ (0.19) \\ 21.29\% \end{array}$	$\begin{array}{c} -0.35 \ (0.75) \\ 1.49 \ (1.03) \\ -3.79 \ (2.05) \\ -1.13 \ (1.31) \\ 13.31\% \end{array}$	$\begin{array}{c} 0.01 \ (0.01) \\ -0.36 \ (0.76) \\ 1.46 \ (1.04) \\ -3.81 \ (2.06) \\ -1.12 \ (1.3) \\ 13.32\% \end{array}$
<i>Results of Mann-W</i> Mean difference	Nutney U test (compar.	ing mean CED of over	seas study (OS)/oversea OS 1.06	s work experience (OW))	MO	- 00	% 18
<i>p</i> -value <i>Results of Mann-</i> ]	Whitney U Test (com,	tþaring mean CED of	0.000 CEO/Chair studied or	verseas (CEOS)/(CHAS)	0.00	0	70
Mean difference <i>p</i> -value			0.118		0.000	39 11	85
Notes: Standard	errors in parenthesis	s. *, ** and ***denote	significance at the 0.	10, 0.05 and 0.01 levels, 1	espectively		
_							
<b>Table</b> Robustness testir							Corporat environmenta disclosure

had a profound positive influence on attitudes towards CED. In addition, Chairs in Vietnam tend to be more influential than CEOs *vis-a-vis* CED, suggesting the power of Chairs in VLFs because of their high ownership, which made them less independent. Overseas work experience, however, was not significant. Our findings underline the value-add of overseas education in forming human capital, which can make positive contributions to sustainability reporting.

Our study makes seven noteworthy contributions to theory and the literature. Firstly, regarding theoretical contributions, Hambrick (2007) argued that a leader's field of vision, selective perception and interpretation are heavily influenced by executives' experiences, values and personalities. We add to UET by examining the source of education of firms' leadership and demonstrating that exposure to overseas education in culturally distant countries does positively alter strategic choice (in this case firm's propensity towards CED disclosures). This finding indicates that leaders' fields of vision, perceptions and interpretations towards environmental issues have been shaped by their overseas experiences. Something that may not otherwise have been at the forefront of their thinking (field of vision) is given priority because of this enlightening educational experience.

A second contribution is that the overwhelming majority of empirical UET studies have used US samples. Hambrick (2007) contends that this may have stacked the deck in favour of significant results given the autonomy possessed by their firm leadership. Our finding shows that background and experiences are also critical for firm leadership in shaping their strategic choices, such as voluntary disclosures on environmental matters. We indicate that even in a regimented system faced by firm leaders in Vietnam, UET still seems to hold.

The third contribution to UET is that the influence on firm leadership's values can be profound in the context of transitional economies such as Vietnam because of exposure to different cultural views, especially those emanating from Anglo countries. Pucheta-Martínez *et al.* (2019), Cui *et al.* (2020), Post *et al.* (2011) and Jackson and Apostolakou (2010) found higher social and environmental commitment in Anglo Cluster nations. It follows that education exposure to these nations is likely to influence Vietnamese views and focus on CED. Moreover, previous studies (Katmon *et al.*, 2017 and Lewis *et al.*, 2014) have been limited to investigating the effects of local qualifications on sustainability reporting.

Fourthly, extending prior studies such as Liao et al. (2020), Xixiong et al. (2018) and Slater and Dixon-Fowler (2009), our study further investigates the cultural traits of board members at multiple levels, including boards in aggregate, as well as comparing the CEOs' influence with that of Chairs on CED levels. Fifthly, in contrast to prior studies (Abeydeera et al., 2016; Ismaeel and Zakaria, 2019; Welbeck, 2017) that found a global standardisation (e.g. GRI) of sustainability reporting, our study draws special attention to the significant impact on CED of exposure to regional cultures through overseas education. Sixthly, we extend prior studies such as Ismaeel and Zakaria (2019), Abeydeera et al. (2016); Khalil and O'Sullivan (2017) and Welbeck (2017), to State-led transition economies, such as Vietnam by reflecting the distinct characteristics of board members' overseas education. Seventhly, our study addressed a major criticism of Hofstede's cultural dimensions by previous studies such as Khlif (2016) that it fails to adjust for multicultural countries, by using 10 regional cultural clusters as defined by House (2004). Cultural clusters were defined by House (2004) based on cultural values, country economic development and linguistic nations across administrative borders. Much prior research into voluntary disclosures tends to rely on Hofstede's cultural dimensions (Khlif, 2016; López-Duarte et al., 2016). We avoided some of the pitfalls of Hofstede through the use of House's (2004) regions.

This study is subject to three major limitations. Firstly, although we controlled industry (Cheng *et al.*, 2017; Muttakin *et al.*, 2016) and firm size (Cuadrado-Ballesteros *et al.*, 2015) in

our research model, our study did not distinguish "not applicable" disclosure item from "non-disclosure" (zero score) when a company had not disclosed that item. Secondly, we did not control for monetary versus non-monetary information in our research model. Thirdly, this study focussed on one transitional country across a relatively small period of four years. This may limit the generalisability of the findings to other countries.

Fruitful avenues for future research are to observe the impacts of CG factors on CED inside the organisation (e.g. board members) and relationships between the inside-outside perspective, for example, the links between BoDs and stakeholders (e.g. the media, nongovernment organisations (NGOs), investors and employees) through qualitative studies that venture into the field and use case studies or interviews (Ismaeel and Zakaria, 2019: Neely et al., 2020; Rao and Tilt, 2016). Interviews with board members would provide detailed insights into the reasons behind low levels of CED in Vietnam and the influences of overseas education on CED. Interviews with other stakeholders, such as investors, NGOs or regulators, could be another avenue for future research to explain their impacts on disclosure behaviours in countries where most investment is locally sourced or sourced from Eastern rather than Anglo nations [24]. Further, our study used traditional media sources (e.g. annual, sustainability and integrated reports and company websites) to capture the quality and quantity of CED. Future studies could include the use of social media means of communication with stakeholders such as Facebook, LinkedIn and Twitter. Moreover, a suggestion for future research of CED is to use the GRI G4 Sector Disclosures Standard, in which an industry-specific set of guidelines for 40 different sectors will be developed up to 2022 (GRI-Sector, 2020).

A final avenue for future research is to investigate whether a large gap exists between environmental activities/performance and environmental disclosures in Vietnam. The association between environmental performance and reporting has attracted considerable debate. Some studies (Al-Tuwaijri *et al.*, 2004; Clarkson *et al.*, 2008; Dawkins and Fraas, 2011; Iatridis, 2013) suggested a positive relationship between corporate environmental performance and CED existed. Others, such as Patten (2002) and Cho *et al.* (2012), argued a negative relationship existed.

This study has valuable implications for policy and practice. The result for CED implies that disclosures to broader stakeholders appear to be an "alien" concept in Vietnam, as accounting reports have been traditionally used for central planning purposes (Vu and Buranatrakul, 2017). This also suggests that sustainability reporting is not effective in transitional economies and needs to be mandatory in countries where low stakeholder orientation and a weak regulatory environment exist, as has been the case in some other developing countries such as India (Gatti *et al.*, 2019). In addition, the insights gained from this study suggest that shareholders should appoint board members who possess international qualifications to enhance sustainability disclosures. Although our research uses Vietnam as an example, the findings of this study may apply to other transitional countries (e.g. China) that share similar cultures or close institutional infrastructures and CG regimes (Mahmood *et al.*, 2019; Situ and Tilt, 2018).

### Notes

 Three similar terms are used in the literature to explain company disclosures concerning the environment: CED is defined as "the set of information items relating to a company's past, current and future environmental management activities and performance" (Berthelot *et al.*, 2003, p. 2). CED has also been called corporate environmental reporting or corporate environmental information disclosure (CEID) (Cheng *et al.*, 2017). Another related term is "CSD", which reflects elements of social and environmental accounting (Deegan, 2016). A final related term is "CSR",

]	MEDAR		which relates to elements, such as corporate responsibility, corporate citizenship, sustainability and corporate social performance (Freeman and Hasnaoui, 2011). For the purposes of our study, we will use CED, as our focus is on environmental disclosures.
		2.	Australia, Canada, England, Ireland, New Zealand, the USA in the Global Leadership and Organisational Behaviour Effectiveness Research Programme (GLOBE) study of 62 societies. GLOBE was funded in October 1993 and recruited GLOBE country co-investigators to collect data (House, 2004).
•		3.	Circular No. 25/2019/TT-BTNMT dated 31 December 2019 of the Ministry of Natural Resources and Environment, available at: https://english.luatvietnam.vn/ircular-no-25-2019-tt-btnmt-dated-december-31-2019-of-the-ministry-of-natural-resources-and-environment-detailing-the-implementation-of-a-number-of-180055-Doc1.html (Accessed: 12 October 2020).
		4.	Decree No. 40/2019/ND-CP dated 13 May 2019 on amending and supplementing a number of articles of the decrees detailing and guiding the implementation of the Law on Environmental Protection, available at: https://english.luatvietnam.vn/ecree-no-40-2019-nd-cp-dated-may-13-2019-of-the-government-on-amending-and-supplementing-to-decrees-guiding-the-implementation-of-the-law-on-enviro-172760-Doc1.html (Accessed: 12 October 2020).
		5.	Decree 155/2016/ND-CP dated 18 November 2016 on penalties for administrative violations against regulations on environmental protection, available at: https://vanbanphapluat.co/decree-155-2016-nd-cp-penalties-administrative-violations-against-regulations-on-environmental-protection (Accessed: 12 October 2020).
		6.	This is a compilation of the <i>National Greenhouse and Energy Reporting Act 2007</i> that shows the text of the law as amended and in force on 30 August 2019.
		7.	The term "qualifications" used in our paper is defined as education or degrees earned by students (e.g. tertiary and higher degrees) based on Brooks <i>et al.</i> (2012) and Parasnis <i>et al.</i> (2008). Given the context of our research was in Vietnam, the terms "overseas qualification" or "foreign qualifications" means foreign education or degrees that board members obtained outside Vietnam, as adopted in Pham (2019b) and Ho <i>et al.</i> (2016).
		8.	China, Hong Kong, South Korea in the GLOBE study (House, 2004).
		9.	The criteria for large firms are based on Regulation of issuing securities listed at the HOSE (Decision No. 10/QD-SGDHCM dated 13 January 2014 by Ministry of Finance (MOF)). A company is defined as "large" for a financial year and is listed at the HOSE if it satisfies at least two of the following criteria, namely, the value of equity at the time of listing registration is at least \$6m (US \$); and the minimum <i>return on equity (ROE)</i> is 5% for the past two years.
		10.	IFC, World Bank Group (IFC), 2013. Sustainability reporting handbook for Vietnamese companies: IFC advisory services in East Asia and the Pacific (English), available at: < <u>http://documents.</u> worldbank.org/curated/en/179691468328537687/Sustainability-reporting-handbook-for-Vietnamese- companies-IFC-advisory-services-in-East-Asia-and-the-Pacific> Accessed 24 October 2016.
		11.	IFC, S 2016, <i>E&amp;S Disclosure Guide</i> , available at: <a href="https://sseinitiative.org/wp-content/uploads/2014/08/20161212_ES-Disclosure-Guideline-ENGLISH.pdf">https://sseinitiative.org/wp-content/uploads/2014/08/20161212_ES-Disclosure-Guideline-ENGLISH.pdf</a> > Accessed 6 April 2017.
		12.	In total, 10 sectors have been classified according to the Global Industry Classification Standard (HOSE, 2016).
		13.	Results showed that only six missing values were found, which fell in only four VLFs in the Industrial sector and these also fell in two missing variables, which were not the main variables in the model. The study applied listwise deletion by Peugh and Enders (2004) to discard these four companies. Furthermore, Enders (2010) pointed out that listwise deletion may only produce biased estimation when deleting cases with a large number of variables, which can lead to a considerable reduction in the total sample size. In our study, only four firms out of the total of 264 were deleted, which accounted for a small proportion (only 1.5%) of the sample size.

14. Six experts (including two reporting specialists from HOSE, one Professor of Accounting from Da Nang University, one Senior Auditor from Deloitte Vietnam and two Senior Consultants from Vietnam MOF agreed that, as CED is voluntary in Vietnam and MOF has not issued any regulations for VLFs to report their environmental performance, the indicators extracted from the GRI framework in the context of Vietnam were optimal in the circumstances.

Corporate environmental disclosures

- 15. For instance, the average level of CED in Vietnam (3.65 out of 90 points) was much lower than that reported in China (10 out of 30 points; Wang *et al.* (2019, p. 10); and 9.7 out of 27 (Cheng *et al.* (2017, p. 216).
- 16. Iran [16 out of 34 (Alipour et al., 2019, p. 590)].
- Nearly half (49%) of VLFs reported environmental information, compared to India, where Garg and Kumar (2018) found that 62% of companies disclosed information regarding environmental agendas.
- A Pseudo R<sup>2</sup> of 25.29% is higher than that reported in Liang *et al.* (2012) (20.8%) and Ben-Amar *et al.* (2017) (20.2%) with acceptable predictability.
- (e.g. Australian Colombo programme, US Pacification Plan, European Region Action Scheme for the Mobility of University Students, Bologna Process and the European Union Commission).
- 20. (e.g. Recommendation 2.5 in Australia by ASX (2019)).
- 21. Vietnamese Law on Enterprises 2014, Article 152, Clause 1, available at: https://thukyluat.vn/vb/ luat-doanh-nghiep-2014-3f692.html (Accessed: 28 June 2020). This states that the Chair can be the CEO or manager except for listed companies with state-ownerships that are more than 50%.
- 22. It is noted that *BMOS/BMOW* were checked for normal distribution and results indicate that these two variables had skewness ( $|\gamma_1| < 3.0$ ) and kurtosis ( $|\gamma_2| < 10.0$ ), which satisfied the thresholds by Kline (2016, p. 76) to perform a Hausman test.
- 23. The first robustness check was sensitivity analysis by changing different variables in the model whilst holding the seven variables of interest (*BMOS, BMOW, BSA, BSC, HIE, BMBA and LAW*) constant (Cameron and Trivedi, 2005; Srivastava *et al.*, 2017). For example, see Models 1 to 3 in Table 6. The second technique used different measurements of variables (e.g. Models 4 and 5). In Model 4, *BMOS* and *BMOW* were measured as dichotomous variables instead of the proportion of board members who studied or worked internationally, which were in the optimal model. This means that *BMOS* or *BMOW* was equal to one if a board member studied or worked overseas, respectively, and zero otherwise. Similar measurements were made for *BSA* and *BSC*. Model 5 did not measure *BMOS, BMOW, BSA or BSC* as a proportion of board members who, respectively: studied internationally, worked internationally, studied in Anglo Cluster regions or studied in Confucian Asia regions all of which were over the total board members of the firm. Rather, these variables were computed as continuous variables based on the number of years that they studied or worked overseas or in a given region. This adjustment was made because the influence of international experience could be greater where the exposure was for a longer period of time (Carpenter *et al.*, 2001; Sambharya, 1996; Sullivan, 1994).
- 24. It is worth noting that in Vietnam, when foreign investment occurred, it was reported to have come from other Asian countries, such as Taiwan, South Korea, Hong Kong, Singapore and China (Hays, 2008; Samuel, 2019). Also, results from our descriptive statistics reveal that a comparatively low percentage of shareholdings were held by foreign investors (5.33%).

#### References

- Abeydeera, S., Tregidga, H. and Kearins, K. (2016), "Sustainability reporting more global than local?", Meditari Accountancy Research, Vol. 24 No. 4, pp. 478-504.
- Acquisti, A., John, L.K. and Loewenstein, G. (2012), "The impact of relative standards on the propensity to disclose", *Journal of Marketing Research*, Vol. 49 No. 2, pp. 160-174.

Adams, H.A. and Do, L.T. (2003), Vietnamese Accounting Standards, ACW, Hanoi.

- Aguilera, R.V., Rupp, D.E., Williams, C.A. and Ganapathi, J. (2007), "Putting the S back in corporate social responsibility: a multilevel theory of social change in organizations", Academy of Management Review, Vol. 32 No. 3, pp. 836-863.
- Akrout, M.M. and Ben Othman, H. (2016), "Environmental disclosure and stock market liquidity: evidence from arab mena emerging markets", *Applied Economics*, Vol. 48 No. 20, pp. 1840-1851.
- Al-Tuwaijri, S.A., Christensen, T.E. and Hughes, K. (2004), "The relations among environmental disclosure, environmental performance, and economic performance: a simultaneous equations approach", *Accounting, Organizations and Society*, Vol. 29 No. 5, pp. 447-471.
- Alipour, M., Ghanbari, M., Jamshidinavid, B. and Taherabadi, A. (2019), "Does board independence moderate the relationship between environmental disclosure quality and performance? Evidence from static and dynamic panel data", *Corporate Governance: The International Journal of Business in Society*, Vol. 19 No. 3, pp. 580-610.
- Ashkanasy, N.M., Trevor-Roberts, E. and Earnshaw, L. (2002), "The anglo cluster: legacy of the British empire", *Journal of World Business*, Vol. 37 No. 1, pp. 28-39.
- Ashwill, M.A. and Thai, N.D. (2005), *Vietnam Today: A Guide to a Nation at a Crossroads*, Intercultural Press, Yarmouth, ME.
- ASX (2019), "Australian corporate governance principles and recommendations", ASX Corporate Governance Council. available at: www.asx.com.au/documents/regulation/cgc-principles-and-recommendations-fourth-edn.pdf (accessed 1 April 2020).
- Auletta, A. (2000), "A retrospective view of the Colombo plan: government policy, departmental administration and overseas students", *Journal of Higher Education Policy and Management*, Vol. 22 No. 1, pp. 47-58.
- Bachmann, R.K.B., Carneiro, L.M. and Espejo, M.M.S.B. (2013), "Environmental information disclosure: a proposed indicator based on experts' perceptions", *Journal of Accounting and Organisations*, Vol. 7 No. 17, pp. 33-44.
- Bamber, L.S., Jiang, J. and Wang, I.Y. (2010), "What's My style? The influence of top managers on voluntary corporate financial disclosure", *The Accounting Review*, Vol. 85 No. 4, pp. 1131-1162.
- Barako, D.G., Hancock, P. and Izan, H.Y. (2006), "Factors influencing voluntary corporate disclosure by kenyan companies", *Corporate Governance: An International Review*, Vol. 14 No. 2, pp. 107-125.
- Barney, J.B. (1992), "Integrating organizational behavior and strategy formulation research: a resource based analysis", Advances in Strategic Management, Vol. 8 No. 1, pp. 39-61.
- Baskerville, R.F. (2003), "Hofstede never studied culture", Accounting, Organizations and Society, Vol. 28 No. 1, pp. 1-14.
- Bear, S., Rahman, N. and Post, C. (2010), "The impact of board diversity and gender composition on corporate social responsibility and firm reputation", *Journal of Business Ethics*, Vol. 97 No. 2, pp. 207-221.
- Bechtel, R.B., Verdugo, V.C. and de Queiroz, J. (1999), "Environmental belief systems: United States, Brazil, and Mexico", *Journal of Cross-Cultural Psychology*, Vol. 30 No. 1, pp. 122-128.
- Becker, T.E., Robertson, M.M. and Vandenberg, R.J. (2018), "Nonlinear transformations in organizational research: possible problems and potential solutions", *Organizational Research Methods*, Vol. 20 No. 2, pp. 1-36.
- Ben-Amar, W., Chang, M. and McIlkenny, P. (2017), "Board gender diversity and corporate response to sustainability initiatives: evidence from the carbon disclosure project", *Journal of Business Ethics*, Vol. 142 No. 2, pp. 369-383.
- Benedict, J.T.K. (2019), Speaking out in Vietnam: Public Political Criticism in a Communist Party–Ruled Nation, Cornell University Press, New York, NY.

Berthelot, S., Cormier, D. and Magnan, M. (2003), "Environmental disclosure research: review and synthesis", <i>Journal of Accounting Literature</i> , Vol. 22, pp. 1-44.	Corporate
Bertrand, M. and Schoar, A. (2003), "Managing with style: the effect of managers on firm policies", <i>The Quarterly Journal of Economics</i> , Vol. 118 No. 4, pp. 1169-1208.	disclosures
Brammer, S. and Pavelin, S. (2006), "Voluntary environmental disclosures by large UK companies", <i>Journal of Business Finance &amp; Accounting</i> , Vol. 33 No. 7, pp. 1168-1188.	
Brochet, F., Miller, G., Naranjo, P. and Yu, G. (2019), "Managers' cultural background and disclosure attributes", <i>The Accounting Review</i> , Vol. 94 No. 3, pp. 57-86.	
Brooks, R., Waters, J. and Pimlott-Wilson, H. (2012), "International education and the employability of Uk students", <i>British Educational Research Journal</i> , Vol. 38 No. 2, pp. 281-298.	
Buchholz, F., Jaeschke, R., Lopatta, K. and Maas, K. (2018), "The use of optimistic tone by narcissistic ceos", <i>Accounting, Auditing and Accountability Journal</i> , Vol. 31 No. 2, pp. 531-562.	
Cameron, A.C. and Trivedi, P.K. (2005), <i>Microeconometrics: Methods and Applications</i> , Cambridge university press, Cambridge.	
Cameron, A.C. and Trivedi, P.K. (2009), "Multinomial models", in <i>Microeconometrics Using Stata</i> , 5th edn, Stata press College Station, TX, pp. 477-518.	
Carpenter, M.A., Sanders, W.G. and Gregersen, H.B. (2001), "Bundling human capital with organizational context: the impact of international assignment experience on multinational firm performance and ceo pay", <i>Academy of Management Journal</i> , Vol. 44 No. 3, pp. 493-511.	
Cheng, Z., Wang, F., Keung, C. and Bai, Y. (2017), "Will corporate political connection influence the environmental information disclosure level? Based on the panel data of a-shares from listed companies in shanghai stock market", <i>Journal of Business Ethics</i> , Vol. 143 No. 1, pp. 209-221.	
Cho, C.H., Guidry, R.P., Hageman, A.M. and Patten, D.M. (2012), "Do actions speak louder than words? An empirical investigation of corporate environmental reputation", <i>Accounting, Organizations</i> <i>and Society</i> , Vol. 37 No. 1, pp. 14-25.	
Chu, T.T.H. (2018), "Environmental pollution in vietnam: challenges in management and protection", Journal of Vietnamese Environment, Vol. 9 No. 1, pp. 1-3.	
CIESIN, Y. (2018), <i>Environmental Performance Index</i> , available at: https://epi.envirocenter.yale.edu/ (accessed 8 April 2019).	
Clark, A., Georgellis, Y. and Sanfey, P. (2001), "Scarring: the psychological impact of past unemployment", <i>Economica</i> , Vol. 68 No. 270, pp. 221-241.	
Clarke, B. and Vu, C.C. (2021), "Eia effectiveness in vietnam: key stakeholder perceptions", <i>Heliyon</i> , Vol. 7 No. 2, pp. 1-11.	
Clarkson, P.M., Li, Y., Richardson, G.D. and Vasvari, F.P. (2008), "Revisiting the relation between environmental performance and environmental disclosure: an empirical analysis", <i>Accounting,</i> <i>Organizations and Society</i> , Vol. 33 Nos 4/5, pp. 303-327.	
Cowen, S.S., Ferreri, L.B. and Parker, L.D. (1987), "The impact of corporate characteristics on social responsibility disclosure: a typology and Frequency-Based analysis", <i>Accounting, Organizations and Society</i> , Vol. 12 No. 2, pp. 111-122.	
Cuadrado-Ballesteros, L., Rodríguez-Ariza, L. and García-Sánchez, IM. (2015), "The role of independent directors at family firms in relation to corporate social responsibility disclosures", <i>International Business Review</i> , Vol. 24 No. 5, pp. 890-901.	
Cui, X., Peng, X., Jia, J. and Wu, D. (2020), "Does board independence affect environmental disclosures by multinational corporations? Moderating effects of national culture", <i>Applied Economics</i> , Vol. 1 No. 2, pp. 1-19.	
Dao, M. and Ofori, G. (2010), "Determinants of firm compliance to environmental laws: a case study of vietnam'", <i>Asia Europe Journal</i> , Vol. 8 No. 1, pp. 91-112.	

MEDAR	Darnall, N., Henriques, I. and Sadorsky, P. (2010), "Adopting proactive environmental strategy: the
	influence of stakeholders and firm size", Journal of Management Studies, Vol. 47 No. 6,
	pp. 1072-1094.

Dawkins, C.E. and Fraas, J.W. (2011), "Erratum to: beyond acclamations and excuses: environmental performance, voluntary environmental disclosure and the role of visibility", *Journal of Business Ethics*, Vol. 99 No. 3, pp. 383-397.

Deegan, C.M. (2012), Australian Financial Accounting, 7th Edn, McGraw-Hill Australia, Sydney.

- Deegan, C. (2016), "Twenty five years of social and environmental accounting research within critical perspectives of accounting: hits, misses and ways forward", *Critical Perspectives on Accounting*, available at: www.sciencedirect.com/science/article/pii/S1045235416300284 (accessed 4 August 2016).
- De Villiers, C. and Marques, A. (2016), "Corporate social responsibility, country-level predispositions, and the consequences of choosing a level of disclosure", *Accounting and Business Research*, Vol. 46 No. 2, pp. 167-195.
- Docquier, F. and Rapoport, H. (2012), "Globalization, brain drain, and development", Journal of Economic Literature, Vol. 50 No. 3, pp. 681-730.
- Elliott, D. (2016), *The Vietnamese War: Revolution and Social Change in the Mekong Delta*, 1930-1975, Routledge, New York, NY.
- Elm, D.R., Kennedy, E.J. and Lawton, L. (2001), "Determinants of moral reasoning: sex role orientation, gender, and academic factors", *Business and Society*, Vol. 40 No. 3, pp. 241-265.
- Enders, C.K. (2010), "Applied missing data analysis", in *Traditional Methods for Dealing with Missing Data*, Guilford press, New York, NY, 37-55.
- Eng, L.L. and Mak, Y.T. (2003), "Corporate governance and voluntary disclosure", Journal of Accounting and Public Policy, Vol. 22 No. 4, pp. 325-345.
- Everaert, P., Bouten, L. and Baele, A. (2019), "Csr website disclosure: the influence of the upper echelons", Accounting, Auditing and Accountability Journal, Vol. 32 No. 2, pp. 421-455.
- Forbes (2016), "Forbes Viet Nam magazine has named the top 50 listed companies of the year in the local stock markets", available at: http://thepangroup.vn/forbes-vn-names-top-50-local-firms-en140666.htm#.WHrp6lN97X4 (accessed 28 December 2016).
- Freeman, I. and Hasnaoui, A. (2011), "The meaning of corporate social responsibility: the vision of four nations", *Journal of Business Ethics*, Vol. 100 No. 3, pp. 419-443.
- Garg, M.C. and Kumar, S. (2018), "The relationship between corporate environmental reporting practices and company characteristics: evidence from India", *Journal of Accounting Research Audit Practices*, Vol. 17 No. 3, pp. 1-16.
- Gatti, L., Vishwanath, B., Seele, P. and Cottier, B. (2019), "Are we moving beyond voluntary csr? Exploring theoretical and managerial implications of mandatory csr resulting from the new Indian companies act", *Journal of Business Ethics*, Vol. 160 No. 4, pp. 961-972.
- Ge, W., Matsumoto, D. and Zhang, J.L. (2011), "Do Cfos have style? An empirical investigation of the effect of individual cfos on accounting practices", *Contemporary Accounting Research*, Vol. 28 No. 4, pp. 1141-1179.
- Ghazali, M.N.A. (2007), "Ownership structure and corporate social responsibility disclosure: some malaysian evidence", *Corporate Governance: The International Journal of Business in Society*, Vol. 7 No. 3, pp. 251-266.
- Giannarakis, G. (2014), "Corporate governance and financial characteristic effects on the extent of corporate social responsibility disclosure", *Social Responsibility Journal*, Vol. 10 No. 4, pp. 569-590.
- Giannetti, M., Liao, G. and Yu, X. (2012), *The Brain Gain of Corporate Boards: A Natural Experiment from China*, Centre for Economic Policy Research London.

Gray,	S.,	Mirke	ovic,	А.	and	Ragu	nathan,	V.	(2006),	"The	deter	minants	of	credit	ratings:	Australian
	ev	idence	e", A1	ıstri	alian	Journ	al of M	ana	gement	, Vol. 3	81 No.	2, pp. 33	3-3	54.		

- GRI-Sector (2020), "Gri G4 sector disclosures", available at: www.globalreporting.org/how-to-use-thegri-standards/questions-and-answers/how-to-use-the-gri-standards/#anchor4 (accessed 20 June 2021).
- GRI (2020), "Gri standards", available at: www.globalreporting.org/how-to-use-the-gri-standards/gristandards-english-language/ (accessed 15 June 2021).
- Gul, F.A. and Leung, S. (2004), "Board leadership, outside directors' expertise and voluntary corporate disclosures", *Journal of Accounting and Public Policy*, Vol. 23 No. 5, pp. 351-379.
- Hagle, T.M. and Mitchell, G. (1992), "Goodness-of-Fit measures for probit and logit", American Journal of Political Science, Vol. 36 No. 3, pp. 762-784.
- Hair, J.F., Ringle, C.M. and Sarstedt, M. (2011), "Pls-Sem: Indeed a silver bullet", *Journal of Marketing Theory and Practice*, Vol. 19 No. 2, pp. 139-152.
- Hambrick, D. (2007), "Upper echelons theory: an update", Academy of Management Review, Vol. 32 No. 2, pp. 334-343.
- Hambrick, D.C. and Mason, P.A. (1984), "Upper echelons: the organization as a reflection of its top managers", *The Academy of Management Review*, Vol. 9 No. 2, pp. 193-206.
- Haque, S. and Deegan, C. (2010), "Corporate climate change-related governance practices and related disclosures: evidence from Australia", *Australian Accounting Review*, Vol. 20 No. 4, pp. 317-333.
- Hays, J. (2008), "Trade, the world trade organization (wto), foreign investment, and foreign companies in Vietnam", *Facts and Details*, Vol. 2019 No. 31.
- Herrmann, P. and Datta, D.K. (2005), "Relationships between top management team characteristics and international diversification: an empirical investigation", *British Journal of Management*, Vol. 16 No. 1, pp. 69-78.
- Ho, S.S. and Wong, K.S. (2001), "A study of the relationship between corporate governance structures and the extent of voluntary disclosure", *Journal of International Accounting, Auditing and Taxation*, Vol. 10 No. 2, pp. 139-156.
- Ho, N.T.T., Seet, P.S. and Jones, J. (2016), "Understanding Re-Expatriation intentions among overseas returnees – an emerging economy perspective", *The International Journal of Human Resource Management*, Vol. 27 No. 17, pp. 1938-1966.
- Hoang, C. (2010), Boat People: Personal Stories from the Vietnamese Exodus 1975-1996, Carina Hoang Communications, Cloverdale, W.A.
- Hoang, T.C., Abeysekera, I. and Ma, S. (2018), "Board diversity and corporate social disclosure: evidence from Vietnam", *Journal of Business Ethics*, Vol. 151 No. 3, pp. 833-852.
- Hoang, T.A., Chu, N.X. and Tran, T.V. (2015), "The environmental pollution in Vietnam: source", Impact and Remedies', Transportation, Vol. 495 No. 112, pp. 38-122.
- Hooks, J. and van Staden, C.J. (2011), "Evaluating environmental disclosures: the relationship between quality and extent measures", *The British Accounting Review*, Vol. 43 No. 3, pp. 200-213.
- HOSE (2016), Ho Chi Minh Stock Exchange, available at: www.hsx.vn/ (accessed 10 July 2016).
- House, R.J. (2004), Culture, Leadership, and Organizations: The Globe Study of 62 Societies, Globe Study of 62 Societies, Sage Publications, Thousand Oaks, CA.
- Hsu, A. (2016), 2016 Environmental Performance Index, CT: Yale University, New Haven, available at: https://web.archive.org/web/20171004102150/http://epi.yale.edu/sites/default/files/2016EPI\_ Full\_Report\_opt.pdf (accessed 19 March 2019).
- Iatridis, G.E. (2013), "Environmental disclosure quality: evidence on environmental performance, corporate governance and value relevance", *Emerging Markets Review*, Vol. 14, pp. 55-75.

- Ismaeel, M. and Zakaria, Z. (2019), "Perception of preparers of sustainability reports in the Middle East", Meditari Accountancy Research, Vol. 28 No. 1, pp. 89-116.
- Ismail, A.M. and Latiff, I.H.M. (2019), "Board diversity and corporate sustainability practices: evidence on environmental, social and governance (esg) reporting", *International Journal of Financial Research*, Vol. 10 No. 3, pp. 31-50.
- Jackson, G. and Apostolakou, A. (2010), "Corporate social responsibility in Western Europe: an institutional mirror or substitute?", *Journal of Business Ethics*, Vol. 94 No. 3, pp. 371-394.
- Jizi, M., Salama, A., Dixon, R. and Stratling, R. (2014), "Corporate governance and corporate social responsibility disclosure: evidence from the Us banking sector", *Journal of Business Ethics*, Vol. 125 No. 4, pp. 601-615.
- Kamal, Y. and Deegan, C. (2013), "Corporate social and environment-related governance disclosure practices in the textile and garment industry: evidence from a developing country", *Australian Accounting Review*, Vol. 23 No. 2, pp. 117-134.
- Karaibrahimoglu, Y.Z. and Cangarli, B.G. (2016), "Do auditing and reporting standards affect firms' ethical behaviours? The moderating role of national culture", *Journal of Business Ethics*, Vol. 139 No. 1, pp. 55-75.
- Katmon, N., Mohamad, Z.Z., Norwani, N.M. and Al Farooque, O. (2017), "Comprehensive board diversity and quality of corporate social responsibility disclosure: evidence from an emerging market", *Journal of Business Ethics*, Vol. 157 No. 2, pp. 1-35.
- Khalil, S. and O'Sullivan, P. (2017), "Corporate social responsibility: internet social and environmental reporting by banks", *Meditari Accountancy Research*, Vol. 25 No. 3, p. 414.
- Khan, A., Muttakin, M. and Siddiqui, J. (2013), "Corporate governance and corporate social responsibility disclosures: evidence from an emerging economy", *Journal of Business Ethics*, Vol. 114 No. 2, pp. 207-223.
- Khlif, H. (2016), "Hofstede's cultural dimensions in accounting research: a review", Meditari Accountancy Research, Vol. 24 No. 4, pp. 545-573.
- Khlif, H., Hussainey, K. and Achek, I. (2015), "The effect of national culture on the association between profitability and corporate social and environmental disclosure", *Meditari Accountancy Research*, Vol. 23 No. 3, pp. 296-321.
- Kim, E.H. and Lyon, T.P. (2011), "Strategic environmental disclosure: evidence from the doe's voluntary greenhouse gas registry", *Journal of Environmental Economics and Management*, Vol. 61 No. 3, pp. 311-326.
- Kipngetich, T.J., Bonuke, R. and Tenai, J. (2019), "Does board education diversity affect environmental accounting disclosure?", SEISENSE Journal of Management, Vol. 2 No. 6, pp. 17-29.
- Kline, R.B. (2016), *Principles and Practice of Structural Equation Modeling*, 4th edn, Guilford Press, New York, NY.
- Lee, K.H. (2017), "Does size matter? Evaluating corporate environmental disclosure in the Australian mining and metal industry: a combined approach of quantity and quality measurement", *Business Strategy and the Environment*, Vol. 26 No. 2, pp. 209-223.
- Lewis, B.W., Walls, J.L. and Dowell, G.W. (2014), "Difference in degrees: ceo characteristics and firm environmental disclosure", *Strategic Management Journal*, Vol. 35 No. 5, pp. 712-722.
- Li, Y., Gong, M., Zhang, X.Y. and Koh, L. (2018), "The impact of environmental, social, and governance disclosure on firm value: the role of ceo power", *The British Accounting Review*, Vol. 50 No. 1, pp. 60-75.
- Liang, J.W., Lin, M.F. and Chin, C.L. (2012), "Does foreign institutional ownership motivate firms in an emerging market to increase voluntary disclosure? Evidence from Taiwan", *Review of Quantitative Finance and Accounting*, Vol. 39 No. 1, pp. 55-76.
- Liao, L., Luo, L. and Tang, Q. (2015), "Gender diversity, board independence, environmental committee and greenhouse gas disclosure", *The British Accounting Review*, Vol. 47 No. 4, pp. 409-424.

- Liao, Z., Huang, Z. and Fang, C. (2020), "Research on the impact of returnee executives' strategic csr orientation on corporate value", *Applied Economics and Finance*, Vol. 7 No. 3, pp. 115-125.
- Logevall, F. (2012), Embers of War: The Fall of an Empire and the Making of America's Vietnam, Random House Publishing Group, New York, NY.
- López-Duarte, C., Vidal-Suárez, M.M. and González-Díaz, B. (2016), "International business and national culture: a literature review and research agenda", *International Journal of Management Reviews*, Vol. 18 No. 4, pp. 397-416.
- Lu, Y. and Abeysekera, I. (2015), "What do stakeholders care about? Investigating corporate social and environmental disclosure in China", *Journal of Business Ethics*, Vol. 144 No. 1, pp. 1-16.
- Lu, Y. and Abeysekera, I. (2017), "What do stakeholders care about? Investigating corporate social and environmental disclosure in China", *Journal of Business Ethics*, Vol. 144 No. 1, pp. 169-184.
- Ma, Y., Zhang, Q., Yin, Q. and Wang, B. (2019), "The influence of top managers on environmental information disclosure: the moderating effect of company's environmental performance", *International Journal of Environmental Research and Public Health*, Vol. 16 No. 7, pp. 1167-1182.
- Mahmood, Z., Kouser, R. and Masud, M.A.K. (2019), "An emerging economy perspective on corporate sustainability reporting – main actors' views on the current state of affairs in Pakistan", Asian Journal of Sustainability and Social Responsibility, Vol. 4 No. 1, pp. 1-31.
- Marshall, R.S. and Brown, D. (2003), "Corporate environmental reporting: what's in a metric?", Business Strategy and the Environment, Vol. 12 No. 2, p. 87.
- Masanet-Llodra, MJ. (2006), "Environmental management accounting: a case study research on innovative strategy", *Journal of Business Ethics*, Vol. 68 No. 4, pp. 393-408.
- Meng, X., Zeng, S., Tam, C.M. and Xu, X. (2013), "Whether top executives' turnover influences environmental responsibility: from the perspective of environmental information disclosure", *Journal of Business Ethics*, Vol. 114 No. 2, pp. 341-353.
- Michalopoulos, C. (1999), Wto Accession for Countries in Transition, The World Bank, New York, NY.
- Muttakin, M.B., Khan, A. and Mihret, D.G. (2016), "The effect of board capital and ceo power on corporate social responsibility disclosures", *Journal of Business Ethics*, available at: https://link. springer.com/content/pdf/10.1007%2Fs10551-016-3105-y.pdf (accessed 27 January 2017).
- Muttakin, M.B., Khan, A. and Mihret, D.G. (2018), "The effect of board capital and ceo power on corporate social responsibility disclosures", *Journal of Business Ethics*, Vol. 150 No. 1, pp. 41-56.
- Neely, B.H., Lovelace, J.B., Cowen, A.P. and Hiller, N.J. (2020), "Metacritiques of upper echelons theory: verdicts and recommendations for future research", *Journal of Management*, Vol. 46 No. 6, pp. 1029-1062.
- Ness, K.E. and Mirza, A.M. (1991), "Corporate social disclosure: a note on a test of agency theory", *The British Accounting Review*, Vol. 23 No. 3, pp. 211-217.
- NGER (2019), "National greenhouse and energy reporting act", available at: www.legislation.gov.au/ Details/C2019C00263 (accessed 12 October 2020).
- Nguyen, C.H. (2014), "Development and brain drain: a review of vietnamese labour export and skilled migration", *Migration and Development*, Vol. 3 No. 2, pp. 181-202.
- Oliver, D.E. and Nguyen, K.D. (2010), "Global competition as a two-edged sword: Vietnam higher education policy", in *Higher Education, Policy, and the Global Competition Phenomenon*, Springer, New York, NY pp. 127-142.
- Orij, R. (2010), "Corporate social disclosures in the context of national cultures and stakeholder theory", *Accounting, Auditing and Accountability Journal*, Vol. 23 No. 7, pp. 868-889.
- Osborne, J. (2010), "Improving your data transformations: applying the box-cox transformation", *Research and Evaluation*, Vol. 15 No. 12, pp. 1-9.

- Papies, D., Ebbes, P. and Van Heerde, H.J. (2017), Addressing Endogeneity in Marketing Models, Springer International Publishing, Cham, 581-627.
- Parasnis, J., Fausten, D. and Cheo, R. (2008), "Do australian qualifications help? The effect of host country qualification on migrant participation and unemployment", *Economic Record*, Vol. 84 No. s1, pp. 131-140.
- Parsons, C. and Vézina, P.L. (2018), "Migrant networks and trade: the vietnamese boat people as a natural experiment", *The Economic Journal*, Vol. 128 No. 612, pp. F210-F234.
- Passetti, E., Cinquini, L. and Tenucci, A. (2018), "Implementing internal environmental management and voluntary environmental disclosure", *Accounting, Auditing and Accountability Journal*, Vol. 31 No. 4, pp. 1145-1173.
- Patten, D.M. (2002), "The relation between environmental performance and environmental disclosure: a research note", Accounting, Organizations and Society, Vol. 27 No. 8, pp. 763-773.
- Peugh, J.L. and Enders, C.K. (2004), "Missing data in educational research: a review of reporting practices and suggestions for improvement", *Review of Educational Research*, Vol. 74 No. 4, pp. 525-556.
- Pham, G.S. (2013), "A political economy approach to the impact of the wto's accession process on Vietnam's economic reform: a case of compliance?", The London School of Economics and Political Science (LSE).
- Pham, L. (2019a), "'International education: a potentiality for ethical development'", in *International Graduates Returning to Vietnam*, Springer, New York, NY, pp. 17-38.
- Pham, L. (2019b), "Introduction: Rethinking international education", *International Graduates Returning to Vietnam, 48, Education in the Asia-Pacific Region: Issues, Concerns and Prospects, Springer, Singapore.*
- Pham, L. (2019c), "Priorities, motivations and expectations of returnees'", in *International Graduates Returning to Vietnam*, Springer, New York, NY, pp. 107-127.
- Plöckinger, M., Aschauer, E., Hiebl, M.R.W. and Rohatschek, R. (2016), "The influence of individual executives on corporate financial reporting: a review and outlook from the perspective of upper echelons theory", *Journal of Accounting Literature*, Vol. 37, pp. 55-75.
- Post, C., Rahman, N. and Rubow, E. (2011), "Green governance: boards of directors' composition and environmental corporate social responsibility", *Business and Society*, Vol. 50 No. 1, pp. 189-223.
- Pucheta-Martínez, M.C., Gallego-Álvarez, I. and Bel-Oms, I. (2019), "Board structures, liberal countries, and developed market economies. Do they matter in environmental reporting? An international outlook", *Business Strategy and the Environment*, Vol. 28 No. 5, pp. 710-723.
- Qi, B., Lin, J.W., Tian, G. and Lewis, H.C.X. (2018), "The impact of top management team characteristics on the choice of earnings management strategies: evidence from China", *Accounting Horizons*, Vol. 32 No. 1, pp. 143-164.
- Radhouane, I., Nekhili, M., Nagati, H. and Paché, G. (2020), "Is voluntary external assurance relevant for the valuation of environmental reporting by firms in environmentally sensitive industries?", sustainability accounting", *Management and Policy Journal (Print)*, Vol. 11, pp. 65-98.
- Rao, K. and Tilt, C. (2016), "Board diversity and csr reporting: an australian study", Meditari Accountancy Research, Vol. 24 No. 2, pp. 182-210.
- Rao, K.K., Tilt, C.A. and Lester, L.H. (2012), "Corporate governance and environmental reporting: an australian study", *Corporate Governance: The International Journal of Business in Society*, Vol. 12 No. 2, pp. 143-163.
- Rest, J.R. and Narváez, D. (1994), Moral Development in the Professions: Psychology and Applied Ethics, L. Erlbaum Associates, Hillsdale, N.J.
- Rivière-Giordano, G., Giordano-Spring, S. and Cho, C.H. (2018), "Does the level of assurance statement on environmental disclosure affect investor assessment?", *Sustainability Accounting, Management and Policy Journal*, Vol. 9 No. 3, pp. 336-360.

- Roodman, D. (2018), "Fitting fully observed recursive mixed-process models with cmp", The Stata Journal: Promoting Communications on Statistics and Stata, Vol. 11 No. 2, pp. 159-206.
- Rosenau, W. (2007), Us Internal Security Assistance to South Vietnam: Insurgency, Subversion and Public Order, Routledge, New York, NY.
- Rupley, K.H., Brown, D. and Marshall, R.S. (2012), "Governance, media and the quality of environmental disclosure", *Journal of Accounting and Public Policy*, Vol. 31 No. 6, pp. 610-640.
- Saggar, R. and Singh, B. (2017), "Corporate governance and risk reporting: Indian evidence", Managerial Auditing Journal, Vol. 32 Nos 4/5, pp. 378-405.
- Sambharya, R.B. (1996), "Research notes and communications: Foreign experience of top management teams and international diversification strategies of Us multinational corporations", *Strategic Management Journal*, Vol. 17 No. 9, pp. 739-746.
- Samuel, P. (2019), "Fdi in vietnam where is the investment going?", Vietnam Briefing, available at: www.vietnam-briefing.com/news/fdi-in-vietnam-investment-by-sector.html/ (accessed 26 July 2019).
- Schwartz, S.H. and Bardi, A. (2001), "Value hierarchies across cultures: taking a similarities perspective", *Journal of Cross-Cultural Psychology*, Vol. 32 No. 3, pp. 268-290.
- Seibert, R.M. and Macagnan, C.B. (2019), "Social responsibility disclosure determinants by philanthropic higher education institutions", *Meditari Accountancy Research*, Vol. 27 No. 2, pp. 258-286.
- Shao-Hui, L. (2009), "The impact of confucianism on Vietnam culture: dissemination and clash", Journal of PLA University of Foreign Languages, Vol. 6, pp. 1-25.
- Shima, K. and Fung, S. (2019), "Voluntary disclosure of environmental performance after regulatory change", *Meditari Accountancy Research*, Vol. 27 No. 2, pp. 287-324.
- Simnett, R., Vanstraelen, A. and Chua, W.F. (2009), "Assurance on sustainability reports: an international comparison", *The Accounting Review*, Vol. 84 No. 3, pp. 937-967.
- Situ, H. and Tilt, C. (2018), "Mandatory? Voluntary? A discussion of corporate environmental disclosure requirements in China", Social and Environmental Accountability Journal, Vol. 38 No. 2, pp. 131-144.
- Slater, D.J. and Dixon-Fowler, H.R. (2009), "Ceo international assignment experience and corporate social performance", *Journal of Business Ethics*, Vol. 89 No. 3, pp. 473-489.
- Srivastava, P., Chen, G. and Harris, A. (2017), "Oral health, dental insurance and dental service use in Australia", *Health Economics*, Vol. 26 No. 1, 1, pp. 35-53.
- Sullivan, D. (1994), "Measuring the degree of internationalization of a firm", Journal of International Business Studies, Vol. 25 No. 2, pp. 325-342.
- Sun, W., Li, X., Geng, Y., Yang, J. and Zhang, Y. (2019), "Board interlocks and the diffusion of csr reporting practices: the role of market development", *Corporate Social Responsibility and Environmental Management*, Vol. 27 No. 3, pp. 1333-1343.
- Sundaramurthy, C., Pukthuanthong, K. and Kor, Y. (2014), "Positive and negative synergies between the ceo's and the corporate board's human and social capital: a study of biotechnology firms", *Strategic Management Journal*, Vol. 35 No. 6, pp. 845-868.
- Udalov, V., Perret, J. and Vasseur, V. (2017), "Environmental motivations behind individuals' energy efficiency investments and daily Energy-Saving behaviour: evidence from Germany, The Netherlands and Belgium", *International Economics and Economic Policy*, Vol. 14 No. 3, pp. 481-499.
- Ullah, S., Zaefarian, G. and Ullah, F. (2020), "How to use instrumental variables in addressing endogeneity? A step-by-Step procedure for non-specialists", *Industrial Marketing Management*, available at: https://doi.org/10.1016/j.indmarman.2020.03.006 (accessed 9 November 2020).

- Unerman, J. (2000), "Methodological Issues Reflections on quantification in corporate social reporting content analysis", Accounting, Auditing and Accountability Journal, Vol. 13 No. 5, pp. 667-681.
- United Nations (2015), "Sustainable development goals", available at: www.un.org/sustainabledevelopment/ sustainable-development-goals/ (accessed 28 December 2019).
- VanDonkelaar, I.S. (2013), "Old english ecologies: environmental readings of Anglo-Saxon texts and culture", PhD thesis, Western Michigan University.
- Vu, K.A. and Buranatrakul, T. (2017), "Examining CSR disclosure in Vietnam: too little, too late", Journal of Business Economics, Vol. 9, pp. 65-79.
- Wang, L. and Juslin, H. (2009), "The impact of chinese culture on corporate social responsibility: the harmony approach", *Journal of Business Ethics*, Vol. 88, pp. 433-451.
- Wang, F., Yang, S., Reisner, A. and Liu, N. (2019), "Does green credit policy work in China?", The Correlation between Green Credit and Corporate Environmental Information Disclosure Quality', Sustainability, Vol. 11, pp. 733-748.
- Weber, R.P. (1990), *Basic Content Analysis*, 2nd ed. Sage University Papers Series. Quantitative Applications in the Social Sciences; 07-049, Sage Publications, Newbury Park, Calif.
- Welbeck, E.E. (2017), "The influence of institutional environment on corporate responsibility disclosures in Ghana", Meditari Accountancy Research, Vol. 25 No. 2, pp. 216-240.
- Welch, A.R. (2010), "Internationalisation of vietnamese higher education: retrospect and prospect", in *Reforming Higher Education in Vietnam*, Springer, New York, NY, pp. 197-213.
- Wiseman, J. (1982), "An evaluation of environmental disclosures made in corporate annual reports", Accounting, Organizations and Society, Vol. 7 No. 1, pp. 53-63.
- Xiao, H. and Yuan, J. (2007), "Ownership structure, board composition and corporate voluntary disclosure evidence from listed companies in China", *Managerial Auditing Journal*, Vol. 226, pp. 604-619.
- Xixiong, X., Hongyan, Z., Weiyu, G. and Wanli, L. (2018), "Returnee executives and improvement of corporate social responsibility performance: an empirical study based on cultural convergence", *Foreign Economics Management Decision*, Vol. 40 No. 5, pp. 99-112.
- Yale University (2014), *Environmental Performance Index*, available at: https://archive.is/ 20140129142747/http://epi.yale.edu/epi/country-rankings (accessed 19 March 2019).
- Yu, EP-y., Guo, C.Q. and Luu, B.V. (2018), "Environmental, social and governance transparency and firm value", *Business Strategy and the Environment*, Vol. 27 No. 7, pp. 987-1004.
- Zeng, S., Xu, X., Yin, H. and Tam, C.M. (2012), "Factors that drive chinese listed companies in voluntary disclosure of environmental information", *Journal of Business Ethics*, Vol. 109 No. 3, pp. 309-321.
- Zhang, J., Kong, D. and Wu, J. (2018), "Doing good business by hiring directors with foreign experience", *Journal of Business Ethics*, Vol. 153 No. 3, pp. 859-876.
- Zhuang, Y., Chang, X. and Lee, Y. (2018), "Board composition and corporate social responsibility performance: evidence from chinese public firms", *Sustainability*, Vol. 10 No. 8, pp. 2752-2765.
- Zrnić, A., Starčević, P.D. and Crnković, B. (2020), "Recent trends in sustainability reporting: literature review and implications for future research", *Review of Contemporary Entrepreneurship*, *Business, Economic Issues*, Vol. 33 No. 1, pp. 271-283.

### **Corresponding author**

Hang Ngoc My Le can be contacted at: lnmhang@hueuni.edu.vn

Арре	endix 1		Corporate
BAGE	$\begin{array}{c} 1.00\\ -0.02\\ -0.03\\ -0.$	mbers of ortion of ortion of ortions with mational fifeations the total as, and 0 ortion of wnership al issued ge of the <i>ontinued</i>	disclosures
CEOS	$\begin{array}{c} 1.00\\ 1.00\\ 0.02\\ 0.03\\ 0.03\\ 0.03\\ 0.02\\ 0.03\\ 0.02\\$	al board me is the prop ooard memi s with inte tional qual transl qual transl power the prop is the prop is the prop SV is leverz (c)	
CHAS	$\begin{array}{c} 1.00\\ -0.15\\ 0.05\\ 0.03\\ 0.03\\ 0.047\\ 0.047\\ 0.04\\ 0.04\\ 0.02\\ 0.04\\ 0.02\\ 0.04\\ 0.02\\ 0.03\\ 0.04\\ 0.01\\ 0.02\\ 0.01\\ 0.01\\ 0.03\\ 0.01\\ 0.03\\ 0.$	wer the tott firm; HIE portion of h in members rith interna prediction of f cortion of f res; BIND ortked; FO ortked; FO orthe State of the State of the State of	
MDIR	$\begin{array}{c} -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.01\\ -0$	ifications of the bers of the bers of the proj n of board members weith multi tive officer issued sha studied/we owned by hm of firm	
BSC	$\begin{array}{c} 1.00\\$	arseas qual ooard mem AW/BACC Proportio n of board 1 members hind execu rise; BGEN rise; BGEN reholding is a logarit is a logarit	
BSA	$\begin{array}{c} -0.04\\ -0.04\\ -0.01\\ -0.01\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.01\\ 0.02\\ 0.02\\ 0.01\\ 0.02\\ 0.02\\ 0.01\\ 0.02\\ 0$	rs with over the total 1 BMBA/BL SSA is the sproportion of 0 other were ther over untries wh arge of sha red; FSIZE	
BACC	$\begin{array}{c} 100\\ 100\\ -0.09\\ 0.00\\ $	if the firm; f the firm; the firm; is BSC is the e properties ted firm, and h board mo index of co the percent h firms list	
BLAW	$\begin{array}{c} 1.00\\ -0.02\\ -0.03\\ -0.03\\ -0.03\\ -0.03\\ 0.02\\ 0.02\\ 0.03\\ 0.02\\ 0.03\\ 0.02\\ 0.03\\ 0.02\\ 0.03\\ $	artion of bo work expe- members of ambers of the firm ADIR is th SEO of a lis ned by each triformance tors; SO is e year which	
BMBA	$\begin{smallmatrix} 1.00\\ 1.00\\ -0.00\\ 0$	is the prop th oversease total board d members of the firm; ed overseas a ( shares ow mmental pe reign inves umber of th	
HIE	$\begin{array}{c} 1.00\\ 0.54\\ 0.54\\ 0.34\\ 0.01\\ 0.01\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.00\\ 0.02\\$	s firm; BMOS members wi tion over the ver the total the total boar at d members airman studi = 1 if a Chair s percentage of V is the envirc ug held by fon I log of the m	
BMOW	$\begin{array}{c} 1.00\\ 0.31\\ 0.32\\ 0.33\\ 0.33\\ 0.33\\ 0.33\\ 0.33\\ 0.33\\ 0.33\\ 0.31\\ 0.04\\ 0.04\\ 0.02\\ 0.02\\ 0.02\\ 0.03\\ 0.01\\ 0.00\\$	of CED for the tion of board higher educa and degrees of the total board the table total the table total age; DUA : BSO is the SSO is the shareholdin is the natura	
BMOS	$\begin{array}{c} 1.00\\ 0.33\\ 0.52\\ 0.52\\ 0.02\\ 0.00\\$	the proport the proport who verseas who verseas up of the second times over egions over eg	
CED	$\begin{array}{c} 1.00\\ 0.006\\ 0.006\\ 0.006\\ 0.007\\ 0.006\\ 0.006\\ 0.006\\ 0.006\\ 0.006\\ 0.006\\ 0.006\\ 0.007\\ 0.006\\ 0.007\\ 0.006\\ 0.007\\ 0.007\\ 0.006\\ 0.007\\ 0.0$	CED is the BMOW is embers with MBA/Law trions in Ar- trions in Ar- trions in Ar- trions in Ar- cican Asia 1 embers of directors ( d by the pr f the compis m; IND is in	
	CED BMOOS BMOOS BMOOS BMDA BMBA BACC BSA BSC BSC BSC BSC BSC BSC BSC BSC BSC BSC	Notes: ( the firm: board m overseas qualifica in Confu board m otherwis the total independ measure shares of listed firm	Table A1.         Pairwise correlations

YEA	
IND	1.00
LEV	0.02
FSIZE	1.00 0.03 0.03
FAGE	00.1 10.0- 10.0-
SO	$^{-0.20}_{-0.21}$
FO	$\begin{array}{c} 1.00\\ -0.02\\ 0.33\\ 0.25\end{array}$
EPW	-0.04 -0.04 -0.02 -0.02 -0.02
EPS	1.00 0.13 0.05 0.06 0.06
BIND	$\begin{array}{c} 1.00\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ \end{array}$
BSO	$\begin{array}{c} -100\\ -100\\ -0.03\\ -0.03\\ -0.03\\ -0.00\\ -0.01\\ -0.01\\ -0.01\\ -0.01\\ \end{array}$
BGEN	$\begin{array}{c} 1.00\\ -0.01\\ -0.01\\ -0.02\\ -0.12\\ -0.12\\ -0.12\\ -0.11\\ -0.01\\ -0.01\end{array}$
DUA	$\begin{array}{c} 1.00\\ -0.06\\ -0.01\\ -0.06\\ -0.06\\ -0.06\\ -0.00\\ 0.00\\ 0.00\\ 0.00\\ 0.00\\ \end{array}$
	CED BMOS BMOW HIE BNBA BLAW BLAW BLAW BAGE BSAC BBAGE BAGE BAGE BAGE BAGE BAGE BBAGA

Table A1.

### Appendix 2

Study overseas Work overseas Board members Anglo countries (%) Board members (%) 86 8.92 35 7.16 Australia Canada 6 0.62 10 2.04 68 England 38 7.05 7.77 Ireland 0 0.00 0 0.00 New Zealand 0 0.00 10 2.04 USA 251 26.04 131 26.79 Total 411 42.63 224 45.80 Field of study 42 4.35 Accounting Finance 138 14.29 MBA 435 45.17 Economics 85 8.84 Engineering 64 6.67 Architect 32 3.27 IT 43 4.49 Chemistry 30 3.13 Law 60 6.26 24 Foreign language 2.45 Other 11 1.09 964 Total 100.00 Table A2. Level of study List of Anglo Diploma 5 0.50 countries where Bachelor's 354 36.77 BoDs studied/worked Certificate after university 0.50 5 overseas and fields/ Master's 478 49.57 levels of overseas PhD 122 12.67 study Total 964 100.00

#### MEDAR Appendix 3

	Code	Environmental items	Obs	Mean	SD	Minimum	Maximum
	EN1 <sup>a</sup>	Materials used by weight or volume	1,040	0.277	0.67	0	3
	EN2	Percentage of materials used that are recycled input materials	1,040	0.160	0.50	0	3
	EN3	Direct energy consumption by primary energy source	1,040	0.347	0.79	0	3
	EN4	Indirect energy consumption by primary source	1,040	0.145	0.51	0	3
	EN5	Energy saved due to conservation and efficiency improvements	1,040	0.338	0.72	0	3
	EN6	Initiatives to provide energy-efficient or renewable energy-based products and services and reductions in energy requirements as a	1,040	0.290	0.68	0	3
	EN7	result of these initiatives Initiatives to reduce indirect energy consumption and achieved	1,040	0.175	0.56	0	3
	ENO	Teductions	1.040	0 1 9 9	0.50	0	2
	ENO	1 otal water withdrawal by source	1,040	0.122	0.30	0	3
	EN9	water sources significantly affected by withdrawal of water (the total	1,040	0.032	0.27	0	3
	EN10	number of significantly affected water sources by type)	1.040	0.117	0.44	0	0
	ENIU	Percentage and total volume of water recycled and reused	1,040	0.117	0.44	0	3
	ENII	Location and size of land owned, leased, managed in of adjacent to, protected areas and areas of high biodiversity value outside protected	1,040	0.027	0.15	0	Z
	EN12	Description of significant impacts of activities, products and services on biodiversity in protected areas and areas of high biodiversity value sutaide protected areas	1,040	0.018	0.13	0	2
	EN12	Unbitate protected or restored	1.040	0.042	0.90	0	2
	EN13 EN14	Strategies, current actions and future plans for managing impacts on biodiversity	1,040 1,040	0.043	0.20	0	$\frac{2}{2}$
	EN15	Number of ILLCN Red List species and national conservation list	1 040	0.012	0.13	0	2
	LIVIO	species with habitats in areas affected by operations, by level of extinction risk	1,040	0.012	0.10	0	2
	EN16	Total direct and indirect greenhouse gas emissions by weight	1.040	0.113	0.46	0	3
	EN17	Other relevant indirect greenhouse gas emissions by weight	1.040	0.057	0.37	Õ	3
	EN18	Initiatives to reduce greenhouse gas emissions and achieved reductions	1,040	0.175	0.52	0	3
	EN19	Emissions of ozone-depleting substances by weight	1,040	0.004	0.07	0	1
	<b>EN20</b>	NOx, SOx and other significant air emissions by type and weight	1,040	0.063	0.38	0	3
	EN21	Total water discharge by quality and destination	1.040	0.257	0.63	0	3
	EN22	Total weight of waste by type and disposal method	1 040	0.235	0.61	Õ	3
	EN23	Total number and volume of significant spills	1,040	0.094	0.48	Ő	3
	EN24	Weight of transported, imported, exported or treated waste deemed hazardous under the terms of the Basel Convention Annex I. II. III and	1,040	0.020	0.23	0	3
	EN25	VIII and percentage of transported waste shipped internationally Identity, size, protected status and biodiversity value of water bodies and related habitats significantly affected by the reporting	1,040	0.005	0.07	0	1
	EN26	organisation's discharges of water and runoff Initiatives to mitigate environmental impacts of products and services	1,040	0.287	0.64	0	3
	EN27	and extent of impact mitigation Percentage of products sold and their packaging materials that are	1,040	0.110	0.45	0	3
	EN28	reclaimed by category Monetary value of significant fines and total number of non-monetary	1,040	0.088	0.36	0	3
		sanctions for non-compliance with environmental laws and regulations					
	EN29	Significant environmental impacts of transporting products and other goods and materials used for the organisation's operations and transporting members of the workforce	1,040	0.085	0.38	0	3
Table 12	EN30 CED	Total environmental protection expenditure and investments by type Sum(EN1:EN30)	1,040	0.084	0.44	0	3
Table A3.							
Details of 30 environmental items	Note: Sourc	<sup>a</sup> EN: Environmental items <b>ce:</b> Definition of each environmental items was retrieved from II	FC and	SSC (2	2013)		